Advanced Fault Diagnosis for Lithium-Ion Battery Systems

Xiaosong Hu 1, Kai Zhang 2, Kailong Liu 2, Xianke Lin 2, Satadru Dey 2, and Simona Onori 2

¹Chongqing University ²Affiliation not available

October 30, 2023

Abstract

Lithium-ion batteries have become the mainstream energy storage solution for many applications, such as electric vehicles and smart grids. However, various faults in a lithium-ion battery system (LIBS) can potentially cause performance degradation and severe safety issues. Developing advanced fault diagnosis technologies is becoming increasingly critical for the safe operation of LIBS. This paper provides a comprehensive review of fault mechanisms, fault features, and fault diagnosis of various faults in LIBS, including internal battery faults, sensor faults, and actuator faults. Future trends in the development of fault diagnosis technologies for a safer battery system are presented and discussed.

Advanced Fault Diagnosis for Lithium-Ion Battery Systems

Abstract: Lithium-ion batteries have become the mainstream energy storage solution for many applications, such as electric vehicles and smart grids. However, various faults in a lithium-ion battery system (LIBS) can potentially cause performance degradation and severe safety issues. Developing advanced fault diagnosis technologies is becoming increasingly critical for the safe operation of LIBS. This paper provides a comprehensive review of fault mechanisms, fault features, and fault diagnosis of various faults in LIBS, including internal battery faults, sensor faults, and actuator faults. Future trends in the development of fault diagnosis technologies for a safer battery system are presented and discussed.

1. Introduction

As one of the most promising energy storage systems, lithium-ion batteries have been widely used in various applications, such as electric vehicles (EVs) and smart grids. Currently, lithium-ion batteries have become the mainstream energy storage solution, owing to their inherent benefits such as high energy density, high power density, and long lifespan. However, the potential risks, due to the abusive operating conditions and harsh environment, pose a huge challenge to the safety of Li-ion battery systems (LIBSs). A real-time effective battery management system (BMS) is critical to ensure the safety of LIBSs. BMS has several functionalities, such as state of charge monitoring, thermal management, charging management, and equalization management. It also tracks the health status and monitors the potential faults of LIBS. Without suitable diagnostics and fault handling, a minor fault could eventually lead to severe damages of LIBS [1]. The importance of the fault diagnostics and fault handling has been demonstrated repeatedly in several severe incidents recently [2]-[4].

There are different fault modes in the LIBS, and fault mechanisms are usually very complex. From a control perspective, these fault modes can be divided into battery fault, sensor fault, and actuator fault. The battery faults, which include overcharge, overdischarge, overheat, external short circuit (ESC), internal short circuit (ISC), electrolyte leakage, battery swelling, battery accelerated degradation, and thermal runaway (TR), are the most critical faults in the LIBS. These faults are also intertwined. Overcharge and overdischarge could lead to various undesirable battery side reactions, resulting in accelerated degradation. These side reactions and gases generated by the chain reactions during TR may eventually cause the battery swelling. Such a swelling along with mechanical damage may, in turn, lead to electrolyte leakage. The ISC is typically caused by separator failure due to manufacturing defects, overheat, mechanical collisions, or penetration by metal dendrites or mechanical punctures. Fortunately, the Joule heat generated by ISC develops into TR only when the equivalent ISC resistance reaches a very low level [5]. Abnormal heat generation occurs under various conditions, such as side reactions during overcharge/over-discharge, ISC, ESC, and contact loose of the cell connector, which further increases battery temperature. Temperature plays an important role in thermal management, battery-pack equalization, capacity/power degradation, and TR [6]. Overheat is the direct cause of the battery TR and can also be facilitated by the chain reactions during TR [7], resulting in a vicious positive feedback cycle. Feng et al. [8] studied the mechanisms of chain reactions during TR for a Li-ion battery with Li(NixCoyMnz)O2 (NCM)/Graphite electrodes and polyethylene-based ceramic coated separator. The solid electrolyte interface (SEI) decomposition, the reaction between electrolyte and anode, the melting of the separator, the decomposition of the NCM cathode and the decomposition of the electrolyte have occurred sequentially in the process of temperature rise. In Ref. [9], they found that 12% of the heat released in the TR of a single cell is sufficient to trigger the TR of the adjacent battery cells. Lamb et al. [10] investigated the failure propagation in a multi-cell Li-ion battery pack when the TR is induced in a single cell. They analyzed the failure propagation under different cell types and electrical connections (parallel or series). Feng et al. [11] summarized four approaches of delaying or preventing TR propagation, including increasing the TR onset temperature, improving heat dissipation, reducing the accumulated energy during TR, and adding thermal resistant layers between adjacent batteries. Hofmann et al. [12] proposed an explosion prevention method by reducing the battery pressure during TR, which is particularly practical for the explosion caused by the electrolyte.

Besides the battery faults, sensor faults can also cause severe issues to the LIBS operation, because all the feedback-based algorithms in the BMS highly depend on the sensor measurements [13]. Sensor faults in LIBS mainly include the voltage sensor fault, current sensor fault, and temperature sensor fault. The current sensor fault affects the accuracy of state of charge (SOC) estimation [14] and multi-state estimation [15], [16]. The estimated SOC and temperature measurements are used to update the battery model parameters in real-time for high-accuracy prediction [17], [18]. Li-ion batteries must be operated within the safe voltage and temperature ranges [19]. Exceeding these ranges may reduce battery performance or even cause accidents. Voltage and temperature sensor faults could also cause equalization errors or thermal management errors in the BMS.

Actuator faults have a more direct impact on control system performance than battery faults and sensor faults. Potential actuator faults in LIBS, including the terminal connector fault, cooling system fault, CAN bus fault, high voltage contactor fault, and fuse fault, are summarized in Ref. [20]. If the cooling system fails, the battery cannot be maintained within the proper operating temperature range, and it may even trigger TR. Battery connection fault will not only cause insufficient power supply but also increase the risk of accidents [21], [22]. Poor connection between batteries leads to a rise in resistance, generates excessive abnormal heat, which further causes temperature rise [23]-[25]. As the charging and discharging process continues, there may be an arc or spark, resulting in the melting of the battery terminals [26].

A lot of research on the fault diagnostics for different components of LIBS has been conducted. Among the different approaches proposed, the most widely used battery fault diagnosis strategy is the model-based approach instead of the data-driven approach because obtaining rich battery fault data is usually time-consuming and costly. In the data-driven approaches, the signal processing methods are mainly used for battery fault diagnosis, rather than machine learning-based methods. Sensor faults and actuator faults usually affect the external signals of the battery, such as voltage, current, and temperature. Therefore, phenomenological models, such as the equivalent circuit model (ECM), are enough for the diagnostic requirements. ECM is simpler in computation and structure than electrochemical models (EM). Therefore, it is easier to design various control and diagnostic tools based on ECM [27]. In general, observer-based methods and signal processing methods are widely used for sensor fault diagnosis and actuator fault diagnosis, respectively.

However, the fault diagnostics for LIBS still faces many challenges. These challenges will be thoroughly discussed in Section 3.4. To the best of our knowledge, there is no review of fault diagnostics for LIBS in the existing literature. For a clear and systematic understanding of the state of the art of LIBS fault diagnostics, this article provides a comprehensive review of fault mechanisms, fault features, and fault diagnosis techniques for Li-ion batteries, sensors, and actuators in the LIBS. The state-of-the-art approaches for LIBS diagnostics and their advantages and limitations are also summarized. In addition, some representative algorithms are classified and discussed to stimulate innovative ideas for LIBS fault diagnosis. Finally, this article discusses future trends and suggestions on improving LIBS fault diagnostics for a safer battery system.

This review is organized as follows. Section 1 introduces the impact of LIBS faults on the safety of LIBS and the relationships between the faults. The research progress of the LIBS fault diagnosis is also briefly summarized. Section 2 introduces the fundamentals of LIBS fault diagnosis, with a focus on methodologies for fault diagnosis systems. Section3 is the main body of this review. It conducts a comprehensive survey on fault mechanisms, fault features, and diagnostic algorithms in LIBS fault diagnosis, and discusses the current problems and challenges in LIBS fault diagnosis. Section 4 presents future trends in LIBS fault diagnosis at three different stages of the fault diagnosis. Section 5 presents the conclusions and suggestions of the latest developments in LIBS fault diagnosis. For a better understanding of the abbreviations used in this review, a list of all acronyms and abbreviations is shown in Table 1.

ABBREVIATIONS						
3σ-MSS	3σ Multi-Level Screening Strategy	NCM	Li(Ni _x Co _y Mn _z)O ₂			
AEKF	Adaptive Extended Kalman Filter	NSMC-EV	National Service and Management Center			
ANN	Artificial Neural Network		of Electric Vehicles			
BMS	Battery Management System	OCV	Open-Circuit Voltage			
CCVC	Charging Cell Voltage Curve	PCBG	Parallel-Connected Battery Group			
ECM	Equivalent Circuit Model	PDE	Partial Differential Equation			
EKF	Extended Kalman Filter	PF	Particle Filter			
EM	Electrochemical Model	PHM	Prognostics and Health Management			
ESC	External Short Circuit	PIO	Proportional Integral Observer			
EV	Electric Vehicle	RCC	Remaining Charging Capacity			
FDI	Fault Detection and Isolation	RF	Random Forest			
FTC	Fault-Tolerant Control	RLS	Recursive Least Squares			
ICA	Incremental Capacity Analysis	SEI	Solid Electrolyte Interface			
ISC	Internal Short Circuit	SOC	State of Charge			
KF	Kalman Filter	SOH	State of Health			
LIBS	Lithium-Ion Battery System	SVM	Support Vector Machine			
MSC	Micro-Short Circuit	TR	Thermal Runaway			
MDM	Mean-Difference Model					

2. Fault-diagnosis systems

Fault diagnosis is a multidisciplinary technology that involves applied mathematics, control theory, information theory, and reliability theory. To have a better understanding of the LIBS fault diagnostics, this section introduces the methodologies of fault diagnosis systems. The basic principles of several representative fault feature extraction and diagnostic algorithms are introduced in detail. Their advantages and will also be discussed. In Section 3, specific fault features and their applications in LIBS fault diagnosis will be elaborated.

2.1. Overview of the fault diagnosis system

The terminology used in the fault diagnosis system is shown in Table 2.

A flowchart of a general fault diagnosis system is shown in Fig. 1. First, the data acquisition is used to process and store data from experimental measurements or high-fidelity simulation models, mainly including voltage, current, and temperature for battery systems. The data processed will be used for model identification, fault characterization, and algorithm verification. Then, key electrical and thermal features are extracted by data pre-processing, feature representation, feature extraction, and feature selection.

Subsequently, fault diagnosis and fault prognosis can be performed based on the feature extracted. The tasks of LIBS fault diagnosis can be divided into fault detection, fault isolation, and fault estimation [28]. Their definitions and differences can be found in Table 2. Fault prognostics can be used to provide early detection or prediction for some battery faults with a slow evolution process. Finally, the fault handling module analyses and evaluates the results from fault diagnosis and fault prognosis, and makes decisions such as alarming, fault-tolerant control, isolating faulty batteries, and even cutting off the power supply.

TABLE 2 - DEFINITION OF THE TERMINOLOGY USED IN FAULT DIAGNOSI	S.
--	----

TERM	DEFINITION			
Fault	An anomaly due to which the system is unable to perform a specified function.			
Fault mode	The macroscopic behavior of fault, also known as the type of fault.			
Fault cause	The key factors causing the fault.			
Fault mechanism	The nature of changes in physical processes, which eventually develop into a fault.			
Fault feature	The feature or parameter that reflects the abnormality caused by a fault.			
Fault detection	The process of determining whether a fault has occurred.			
Fault isolation	The process of determining the type and/or location of the fault.			
Fault identification or estimation	The process of determining the magnitude/intensity of the fault.			
Fault diagnosis	The whole process of detection, isolation, and estimation of the fault.			
Data acquisition	Fault diagnosis			



FIGURE 1 - Flowchart of the general fault diagnosis system. Adapted from [29], [30].

2.2. Feature extraction

Feature extraction is a pre-processing step for fault diagnostics. The accuracy of feature extraction highly depends on the method used. Here, we focus on two main feature extraction methods: signal processing based and model-based.

Various signal processing techniques have been developed to extract useful features in the time, frequency and time-frequency domains, such as root mean square amplitude, spectral analysis [31], wavelet transformation [32], entropy-based method [33], rough set [34] and principal component analysis [35]. For example, battery fault and connection fault can cause abnormal fluctuations in battery voltage response and temperature response. The entropy-based method [33] can be used to capture these anomalies due to its capability of measuring the degree of randomness or disorder of time series data. Note that some methods, such as rough set [35], and principal component analysis [35], can reduce the dimension of the fault features, which is very useful for reducing the complexity of the diagnostic system.

TABLE 3 - COMPARISON OF TWO FEATURE EXTRACTION METHODS.							
FEATURE	KEY TECHNOLOGIES	ADVANTAGES	DISADVANTAGES				
EXTRACTION							
METHODS							
Signal processing-	Analyze the measured data directly to determine the fault.	No need for system modeling;	Difficult to detect minor and				
based method		Easy to analyze qualitatively;	unforeseen faults, and achieve				
		Various applications.	fault isolation.				
Model-based	Use state estimation or parameter estimation algorithms	Easy to implement quantitative	Affected by model accuracy.				
method	to identify changes in system state or model parameters.	analysis and fault isolation.					

Based on the measurements, model-based state estimation and parameter estimation methods can be used to extract fault features. In battery systems, fault features can be characterized by changes in battery states or model parameters. For example, a connection fault can cause a significant change in battery contact resistance. ISC fault and thermal fault can be characterized by the SOC decrease and ohmic internal resistance increase, respectively. For different fault models, the filter [17], observer [18] or least-squares algorithm [36] need to be designed accordingly to extract key states or parameters. Theoretically, artificial intelligence algorithms can also be applied to extract fault features as an alternative to the physics-based model. This method is expected to extract more accurate features with online training and continuous improvement, but with the computational cost of continuous training.

For battery system faults, the advantages and disadvantages of the above two types of feature extraction methods are shown in Table 3. Many battery system faults can cause the capacity fade, extra charge depletion, increased heat generation, or increased battery cell inconsistency. This abnormal behavior can be captured by analyzing the external voltage and temperature response of the battery system using signal processing methods. The signal processing method does not require modeling work, but it may not achieve fault isolation as some of the battery faults have similar electrical and thermal responses. Besides, the signal processing method can only detect the faults when the abnormality in the battery system response reaches a certain level, which makes it difficult to detect minor faults. In contrast, it is easier for the model-based method to quantify and locate specific faults by exploiting the relationship between faults and model states or parameters.

2.3. Diagnostic methods

There are many studies on diagnostic methods. As shown in Fig. 2, we classify the diagnostic methods into the knowledge-based, model-based, and data-driven ones, according to the Refs. [19], [37]-[40].



FIGURE 2 - Classification of fault diagnostic methods.

2.3.1. Knowledge-based methods. These diagnostic methods utilize the knowledge or observation of battery systems and are especially suitable for the nonlinear and complicated systems such as LIBS without the need for developing mathematical models. Although their working principles and diagnostic results are easy to interpret, further studies on LIBS fault mechanisms, knowledge acquisition, and knowledge representation are still required when they are applied to LIBS fault diagnosis. The most widely used knowledge-based methods include graph theory-based, expert system, and fuzzy logic-based.

Specifically, by using graph theory, such as signed directed graph [41], fault tree [42], and failure mode and effects analysis [43], a fault diagnosis network can be constructed based on the fault propagation relationship between various components in the system. Then, a fault can be located using the relevant search theory. An expert system is a designed computer program to simulate the reasoning and decision making of human experts [44]. The knowledge and rules are established by utilizing the historical database and the rich experience from domain experts. Fuzzy logic which conforms to the natural thinking process of human beings and facilitates the processing of qualitative knowledge can be applied to fault diagnosis by using fuzzy parameters, fuzzy models, or fuzzy thresholds.

TABLE 4 - COMPARISON OF KNOWLEDGE-BASED DIAGNOSTIC METHODS.					
KNOWLEDGE-	KEY TECHNOLOGIES	ADVANTAGES	DISADVANTAGES		
BASED METHODS					
Graph theory	Diagnostic network;	Clear causality;	Needs a thorough understanding of fault		
	Fault propagation relationship;	Easy to interpret the diagnosis results;	mechanism;		
	Search strategy.	Easy to analyze qualitatively.	Not suitable for systems with high complexity.		
Expert system	Knowledge acquisition and	No need for the mathematical model;	Difficulty in knowledge acquisition and		
	representation;	Diagnostic results are easy to understand.	representation;		
	Knowledgebase;		Over-reliance on the representativeness and		
	Rule base.		integrity of knowledge.		
Fuzzy logic	Fuzzy rules;	Suitable for handling qualitative	No self-learning ability;		
	Membership function.	knowledge and reasoning.	Difficult to develop effective rules.		

Table 4 gives a comparison of various knowledge-based diagnostic methods in terms of their key technologies, advantages, and disadvantages. Multiple battery faults, sensor faults, and actuator faults may occur in the battery system. Graph theory has a clear causal relationship, and diagnostic results are easy to interpret. However, the complex fault mechanisms of the battery system make it difficult to establish an accurate diagnostic network. The expert system method does not require a physics-based model. However, there also exist several problems when it's applied to battery systems, such as difficulties in knowledge acquisition, inaccurate knowledge representation. The fault states of batteries can be characterized by the anomalies such as rapid SOC decline, intense heat generation, and large voltage fluctuations. These fuzzy parameters can be processed by the fuzzy logic method. However, developing effective rules is still a big challenge.

2.3.2. Model-based methods. For the model-based fault diagnostics, a residual signal is typically obtained by comparing the measurable signal with the signal generated by the model [45]. Subsequently, the residual will be evaluated to determine the diagnostic results [46]. The development of high-fidelity battery models [47], including electrical models, thermal models, and multiphysics models, provides the basis for the model-based fault diagnosis. Thanks to the in-depth understanding of battery system dynamics, these methods can not only detect faults but also locate faults and estimate their magnitude. Therefore, they are becoming the mainstream method for LIBS fault diagnostics. It should be noted that these methods could be affected by model uncertainty, interference, and noise. Model-based methods can be divided into four categories, including the state estimation, parameter estimation, parity space, and structural analysis theory.

MODEL-BASED METHODS	KEY TECHNOLOGIES	ADVANTAGES	DISADVANTAGES
State estimation	Reconstruct system state with filters or observers.	Good real-time performance; No need for a large number of inputs stimulus.	Difficult to determine the fault location and damage degree.
Parameter estimation	Estimate system parameter or fault parameter.	Conducive to fault isolation.	Requires high precision of modeling and sufficient input excitations.
Parity space	Equivalent relationship between input and output variables expressed by the system model.	Simple, fast; Suitable for fault isolation.	Affected by model accuracy and noise.
Structural analysis theory	Structural analysis of system dynamic equations.	Easy to analyze fault detectability and isolability; Workload of selecting residual generators is reduced.	Strongly dependent on the redundant information of the system model.

TABLE 5 - COMPARISON OF MODEL-BASED DIAGNOSTIC METHODS.

The state estimation methods essentially utilize an observer or filter to reconstruct or estimate the internal states, such as SOC and the internal temperature of batteries. After that, the residuals containing the fault information can be obtained by comparing the estimated signals with the sensor measurements [37]. The basic idea of the parameter estimation for fault diagnosis is that faults will affect the physical system process, further leading to the change in model parameters [48], [49]. Therefore, fault detection and isolation (FDI) of LIBS can be achieved by detecting changes in the battery's electrical model and thermal model parameters. The

dynamic model of the battery system determines the relationship between input and output variables. The parity space method can be used to verify this relationship by analyzing the input and output measurements of the battery system [50], [51]. Structural analysis theory finds and utilizes the structural over-determined part of system dynamic equations [52], and then achieves the structural detectability and isolability analysis of faults [53]-[58].

A comparison of the above model-based methods is illustrated in Table 5. Various filters and observers have been applied to fault diagnosis for LIBS, such as Kalman filter (KF) [59], extended Kalman filter (EKF) [60], unscented Kalman filter [61], particle filter (PF) [62], Lunberger observer [63], and adaptive observer [64]. The state estimation method can help the state monitoring function of BMS and can detect the fault with excellent real-time performance. In comparison, parameter estimation methods such as filter methods [65] and least-squares methods [66], can be combined with other methods to locate specific LIBS faults. However, they require higher battery model accuracy and sufficient current excitations [67]. For the parity space methods, such as the parity equation method [68] and the constrained optimization method [69], fault isolation of sensors and actuators in LIBS can be easily achieved based on the different hypothesized no-faulty subsets of inputs and outputs. One obvious advantage of structural analysis theory is the ability to provide fault detectability and isolability analysis regardless of the LIBS parameter values, which greatly reduces the workload of designing residual generation for fault isolation.

2.3.3. Data-driven methods. These methods analyze and process the running data directly to detect faults without relying on the accurate analytical model and the experience of experts. For the data-driven fault diagnosis of LIBS, the fault detection process is simplified by not considering the complicated fault mechanism and system structure, especially for TR and battery accelerated degradation which are affected by various unclear and coupled factors. However, the implementation of this method generally requires a proper pre-processing of raw data for LIBS. Due to the neglect of fault mechanisms, it is not easy to analyze and interpret faults using this method. Furthermore, some data-driven methods also present inherent limitations, such as the need for a large amount of historical data, accompanied by high computational cost, and training complexity [70]. The data-driven methods commonly used in fault diagnosis domain include signal processing, machine learning, and information fusion.

Fault diagnosis based on signal processing usually uses various signal processing techniques to extract fault feature parameters such as deviation, variance, entropy, and correlation coefficient. After that, the fault will be detected by comparing them with the values in a normal state. The artificial neural networks (ANN) [71] and support vector machines (SVM) [72] are two typical machine learning algorithms. ANN-based fault diagnosis is to learn the implicit rules from a given pair of inputs and outputs in the offline training phase, and then to form a nonlinear black-box model for use in the online operation phase. The well-trained ANN can distinguish between the normal and abnormal states of the battery system. The main function of SVM-based fault diagnosis is to transform the input space into a high-dimensional space through a kernel function and to find the optimal hyperplane in this new space. This method treats LIBS fault diagnosis as a sample classification problem and trains an accurate classifier based on historical data. Information fusion represents a process of reasoning and decision-making based on uncertain information. Based on the analysis of multi-source information, more reliable fault detections can be achieved.

Table 6 illustrates a comparative analysis of these data-driven diagnostic methods. Due to the neglect of the LIBS dynamics, the signal processing method is easy to implement and suitable for fault detection, but it is difficult to locate faults directly in the case of multiple LIBS fault coupling. Machine learning algorithms have the learning ability to adapt the training sample set by adjusting its own parameters, and the ability to extract knowledge from the current training samples. Theoretically, the battery black-box model based on the ANN can achieve higher accuracy than the EM and ECM of battery. However, the lack of LIBS fault data may cause over-fitting problems. That is, ANNs with poor generalization ability are likely to cause an undesired false alarm of the LIBS fault. Compared with ANN, SVM has better generalization ability and is applicable to the small sample cases [73], which is especially suitable for the LIBS with a limited amount of fault data. The most critical issue of SVM is the optimal kernel function selection for a specific problem. In order to make full use of existing multi-source information of the LIBS to improve the accuracy of fault diagnosis, an effective fusion algorithm is essential.

DATA-DRIVEN METHODS	KEY TECHNOLOGIES	ADVANTAGES	DISADVANTAGES
Signal processing	Appropriate signal processing techniques.	Easy to implement; Applicable to both linear and nonlinear systems.	Difficult to detect minor faults and directly locate faults; Not suitable for systems with highly coupled components.
Artificial neural network	Neural network structure; Dynamic adjustment of variable weights.	Self-learning from samples; Strong adaptability; Parallel processing.	Need massive historical data and long training time; Poor generalization ability; Over-fitting problems.
Support vector machine	Kernel function selection.	Good generalization ability; Applicable to small sample cases.	Difficult to select the optimal kernel function; Low efficiency for large-scale training sets.
Information fusion	Appropriate information fusion algorithms.	More accurate diagnostic result.	Difficult to select effective fusion algorithms.

TABLE 6 - COMPARISON OF DATA-DRIVEN DIAGNOSTIC METHODS.

2.4. Fault-tolerant control

Fault-tolerant control (FTC) is used to maintain safe operation and meet certain performance requirements when the fault occurs in the system [74]. An architecture of FTC is shown in Fig. 3. In general, FTC can be classified into active FTC and passive FTC [75]. There are few existing studies on FTC in LIBS, and the passive FTC is used in most cases.

The purpose of passive FTC is to design a robust controller such that the system is robust to certain faults [76]. Passive FTC assumes prior knowledge of faults, and therefore, does not need to know the real-time fault information or adjust the controller online. However, passive FTC may be ineffective under unknown faults. Hu et al. [77] developed a dual-redundancy method to achieve FTC of temperature sensors. When a sensor fault occurred, the optimal value determined by relevant algorithms was taken as the sensor output value to ensure proper operation of the system. Berdichevsky et al. [78] introduced that each cell was equipped with two fuses for the cell's anode and cathode in the Tesla Roadster battery pack. This scheme mainly relies on the structural design of the battery system, and can effectively prevent the entire battery system from malfunctioning in the case of a short circuit. However, the requirement of substantial additional components would make the battery structure complicated.

In contrast, based on the real-time information of fault, active FTC will re-adjust the controller parameters or even change the configuration after the fault occurs. That is, active FTC can actively process the fault in real-time so that the system can still achieve the specified functions under fault conditions [79]. Despite high complexity and high computational cost, this method improves the system performance greatly, which has attracted increasing attention in both academia and industry. Therefore, the application of active FTC in LIBS has the potential to become an important research field.



FIGURE 3 - The architecture of fault-tolerant control. Adapted from [80].

2.5. Evaluation system

For battery system faults, the performance of the diagnostic system will vary based on different diagnostic methods. A good evaluation system can compare various diagnostic algorithms and help design a better fault diagnosis method. The key to establishing a good evaluation system for fault diagnosis is to establish a reasonable performance index system and develop appropriate evaluation methods [81].

According to different functionalities, the major performance indexes for the diagnostic system can be roughly divided into detection performance, diagnostic performance, and robustness [82], [83]. Detection performance can be assessed by sensitivity, time delay, false alarm rate, missed detection rate, and misclassification rate. This index is closely related to the timeliness of fault handling for LIBS. Diagnostic performance refers to the capability of fault isolation and the accuracy of fault estimation. False alarms and missed detections are common indices of LIBS, which can cause additional troubleshooting and safety risks. Robustness is the most difficult performance index to measure and achieve. A diagnostic algorithm without the robustness to model uncertainty, interference, and noise can be hardly used in practical LIBS.

To date, there are no standardized evaluation methods for LIBS fault diagnosis. In general, an evaluation method of the diagnostic system follows the process of determining the weight of each index, evaluating each index, and determining the final evaluation result. The weight has a big impact on the final evaluation results. Typically, the weight is related to the importance and reliability of indexes. For example, a battery short-circuits fault has a higher weight than a sensor fault owing to its higher threat to the LIBS. In most cases, it is difficult to evaluate some indicators quantitatively, for example, robustness. One possible solution is to first qualitatively evaluate each index and then quantify the qualitative index in a unified framework.

3. Fault diagnosis for LIBS

Fault diagnosis is critical to ensure the safety of LIBSs. Therefore, it is necessary to study the fault mechanisms, fault features, and diagnostic methods for LIBSs. Fig. 4 illustrates an overview of the faults in LIBS. The faults of LIBS are affected by inherent

defects, improper use, and harsh environments. Therefore, these internal and external factors and their complicated coupling relationship make fault diagnosis of LIBS a difficult task. In general, the faults of LIBS are hidden, and it is difficult to directly and accurately determine early fault conditions by voltage, current, and temperature signals only. Each type of faults poses a certain threat to the LIBS. Battery faults could lead to system performance degradation and even catastrophic accidents such as battery burning and explosion. Sensor faults of BMS will affect the normal operation of the control system, leading to ineffective state estimation, equalization management, and thermal management in the LIBS. Actuator faults often lead to ineffective control actions, which further affects the system response.

3.1. Li-ion battery fault diagnosis

3.1.1. Li-ion battery fault mechanisms. The studies on the battery fault mechanisms provide useful insights into the battery failure process. The understanding of fault mechanisms serves as a foundation for developing the fault diagnostic methods. Currently, a few review papers have been published on the fault mechanisms of Li-ion batteries [84]-[87]. At the microscopic level, Alavi et al. [27] summarized several electrochemical failures such as the loss of electrical contact, current collector corrosion, SEI growth, electrolyte decomposition, fracture in the lattice structure of electrodes, lithium plating, loss of active material, negative electrode diffusion coefficient reduction, porosity change of the electrode, and change of particle size. In addition, several battery faults, including the overcharge/overdischarge, battery accelerated degradation, battery swelling, electrolyte leakage, ESC, ISC, overheat, and TR, are very important in the real applications.

Although the cut-off voltage can be pre-set in the protection circuit, overcharge and overdischarge faults still occur in EVs due to the inconsistency among cells, inaccurate condition monitoring, and charging system faults [88]. For example, if the voltages of series cells are not monitored well in BMS, the cells that have the highest and lowest voltages will be overcharged and overdischarged, respectively, resulting in the rapid aging of the battery.



FIGURE 4 - Overview of the faults in the Li-ion battery systems.

Accelerated battery degradation is caused by the undesired side reactions within the cell, which are accompanied by the losses of cyclable Li-ions and active material [89], [90]. Typically, these adverse side reactions, such as the phase change or decomposition of cathode material [91], electrolyte decomposition [89], SEI decomposition and growth at the anode [92], [93], are caused by various external factors [38] including overcharge/overdischarge, low temperature, high voltage storage, high-rate cycling. Specifically, the

excessive delithiation of anode causes SEI decomposition during overdischarging [94]. After recharging, the newly regenerated SEI changes the electrochemical properties of the anode [95], resulting in an increase of resistance and the degradation of capacity [96]. Repeated overdischarge will accelerate battery capacity degradation, the extent of which depends on the depth of discharge [88]. During overcharging, lithium deposition (mossy or dendritic type) will occur at the surface of the anode [97]. Meantime, over de-intercalation of lithium will contribute to irreversible phase change and even collapse of cathode structure with gas release and heat generation. Temperature is also a very important factor affecting battery operation. Under low-temperature charging conditions, lithium plating is more likely to occur at the anode due to the slow diffusion process [98]. High temperatures can cause SEI decomposition and accelerate the capacity fade. The battery capacity can drop significantly when it is operated or stored above 50 °C, especially in the high SOC range [6]. Under high-rate discharge conditions, a large amount of Li-ion is transferred in a short time, which may cause incomplete de-intercalation of Li-ions and capacity loss.

Battery swelling, electrolyte leakage, and ESC faults are often caused by other battery faults or component faults. There exists a causal relationship between them. Firstly, the gas generated by the side reactions during overcharge and the chain reactions during TR may cause the internal pressure to rise, and even explosion. Then, the battery swelling and mechanical damage become the main causes of electrolyte leakage. Finally, electrolyte leakage can further cause the battery ESC and also the short circuit of adjacent electronic components. Besides, EVs may suffer from water immersion, collision deformation, electric wire failure during operation. Therefore, an ESC may also occur when the electrodes with voltage differences are accidentally connected by conductors [99]. The ESC is a fast discharge process and results in abnormally high heat generation.

ISC, one of the most common faults in TR, can be caused by different separator failures such as deformation, penetration, shrinkage, or melting. For example, mechanical loading [100] can cause the deformation and fracture of the separator, and the electrical short-circuit under mechanical loading is generally predicated by the formation of internal cracks in the battery stack [101]. Separator penetration can be caused by mechanical shock or dendrite due to overcharge and overdischarge [62]. Moreover, the thermal runaway reaction is more severe when the penetration occurs at the center of the battery [102]. The separator shrinkage or melting caused by high temperature and the contamination of the separator by impurities are also the causes of separator failure [103]. Once the separator fails, the ISC is triggered by the contact between anode and cathode. Battery capacity [104] and the heat accumulated during the initial phase [105] are key factors in determining the consequences of ISCs. Studies [106] show that the worst location for ISC is the edge of the electrode where the heat dissipation is limited by the low thermal conductivity of electrolyte and separator materials.

Overheat and TR have mutually reinforcing relationships. The causes of TR are summarized in Ref. [99], including mechanical abuse, electrical abuse, and thermal abuse. Specifically, thermal abuse or overheat is the direct cause of TR. Overheat is usually caused by abnormal heat generation, external heat transfer, and poor heat dissipation. Abnormal heat generation occurs in many scenarios, such as side reactions during overcharge and overdischarge, ESC, ISC, and battery connection fault. A portion of the heat could be transferred to adjacent cells and the environment. Moreover, fault or improper design of the cooling system can also result in poor heat dissipation. Battery overheat caused by these factors may trigger TR, and the mechanism of chain reactions during TR for a Li-ion battery is elaborated in Ref. [107], including capacity degradation at the high temperature, SEI decomposition, the reaction between anode and electrolyte, separator melting, cathode decomposition, electrolyte decomposition, the reaction between anode and binder, electrolyte burning, etc.

3.1.2. Li-ion battery fault features. It is important to note that, unlike sensor fault and actuator fault, data acquisition is a key step in battery fault diagnosis. Therefore, the state of the art of data acquisition for battery fault diagnosis is discussed in this subsection before the Li-ion battery fault feature. Besides the real-time data from EVs, data acquisition can also be achieved through substitute tests and simulation models in academic research.

Many test methods have been developed for Li-ion battery research, such as penetration [106], mechanical loadings [108], [109], external heating [110], overcharge [111], and ESC test [112]. The implementation of the aforementioned test methods often requires a combination of advanced techniques including optical, infrared, chemical, and thermal methods [111], [113]-[115]. For ISC fault, the most concerned battery fault, the currently accepted ISC substitute tests mainly include penetrating, adding phase change material into the separator, inducing the dendrite growth by electrical abuse, and connecting an equivalent ISC resistance in parallel to the cell. Moreover, a new approach to conduct the ISC substitute test is provided by controlling the separator porosity and the pressing force [116]. Because the experimental test methods are costly and time-consuming, a lot of research has been devoted to developing a high-fidelity model that can simulate the battery failure behavior, such as equivalent circuit model [60], [117], two-state thermal model [118], electrochemical-thermal model [119], [120], 3D electrochemical-thermal model [105], 3D electrochemical-thermal ISC coupled model [66], [121], mechanical-electrical-thermal coupled model [122], and finite element model [123].

In general, the Li-ion battery fault feature can be obtained from two sources. First, the battery fault feature can be directly extracted from measurements or transformed from basic features. Second, the fault feature of the battery can also be reflected by certain model parameters. In general, battery faults are typically difficult to be determined by current, voltage, and temperature measurements. Instead, fault features are often extracted from the abnormal responses caused by faults through signal processing. For example, due to the extra charge depletion, ISC can be inferred by two implicit features, including the continuous reduction of the SOC and the rising of the heat generation [124]. These two features can be captured by the responses of battery voltage and temperature [125]. For the short-circuit under mechanical abuse conditions, a local force drop [126] can be regarded as a fault feature of short-circuit, which is consistent with the voltage drop and temperature rise. Moreover, fault features transformed from basic features allow

detecting battery faults more sensitively. For example, in Ref. [117], the differential of the voltage and the fluctuation function of the internal resistance are considered as the fault features. In Ref. [125], the correlation coefficient between cell voltages can capture the abnormal voltage drop. The entropy of battery temperature [127] and voltage [128] become the features of temperature abnormity and voltage fault, respectively.

For the quantitative analysis of faults, certain parameters of the battery model are regarded as fault features, such as the ISC equivalent resistance [129] and thermal model parameters [63], [130] related to convective cooling resistance fault, internal thermal resistance fault and TR fault. Liu [131] and Wu [132] analyzed the relationship between battery faults and parameter changes, and summarized the diagnostic rules for common battery faults. In many studies, certain model parameters are regarded as the state of health (SOH) indicator, such as capacity and internal resistance [133]-[135]. However, the results derived from these parameters may vary at different operating conditions [136]. Since certain electrochemical properties are uniquely related to the degree of battery degradation regardless of operating conditions, they can be used as an indicator for battery SOH, such as the side reaction current density [137]. It should be noted that most battery fault features are at the cell level. In the case of series-connected battery modules, the difference of SOC as well as ohmic internal resistance [36], [138] in the mean-difference model (MDM) of the battery pack can be used as effective ISC fault features.

3.1.3. Li-ion battery fault diagnosis methods. Due to the lack of internal information and strong coupling among various battery faults, many conventional diagnostic methods applied in other fields are not suitable for battery fault diagnosis. Currently, methods used in battery fault diagnosis mainly consist of model-based, data-driven based, knowledge-based, and methods of integrating multiple diagnostic techniques. A comparison of battery fault diagnosis methods is shown in Table 7.

Based on the battery model and measured data, model-based methods use the state estimation and parameter estimation techniques to generate residuals and detect faults. Fault isolation can be achieved by constructing a fault signature table. Due to its simplicity and intuitive nature, the model-based method is widely used for fault diagnosis in battery cells and packs [139]. Based on the electrochemical model, Alavi et al. [62] estimated the transport rate of Li-ions in both positive and negative electrodes by the PF algorithm and then compared the estimated data with the boundary condition to detect the lithium plating. Because overcharge and overdischarge can cause the model parameters to change, Sidhu et al. [59], [60] constructed multiple battery signature-fault models by impedance spectroscopy technology and equivalent circuit methodology, using KF or EKF to estimate the model terminal voltage and generate residuals. A probability-based approach was also applied to indicate the probability of failure. But this method is accompanied by the difficulty of identifying multiple models and running EKF. Dey et al. [63] added convective cooling resistance fault, internal thermal resistance fault, and TR fault into a two-state thermal model, and the FDI is designed for these three thermal faults based on the Luenberger observer. In Ref. [130], Dey et al. extended the previous work to consider the temperature dependence of internal resistance, and incorporate a Lyapunov-based nonlinear observer approach to deal with the nonlinearities of battery. Moreover, in Ref.[140], they utilized a partial differential equation (PDE) model-based scheme and realized the detection and estimation of the size of the thermal fault. Since the excessive charge depletion and abnormal heat generation of the ISC cells affect the voltage and temperature responses, the correlation can also be captured by the phenomenological model. Feng et al. [66] estimated model parameters using the recursive least squares (RLS) with a forgetting factor. Their model includes parallel resistance, capacitance, ohmic resistance, and the temperature derivative of the equilibrium potential in the energy conservation equation. Then, the ISC fault detection was implemented based on the changes in these key parameters. Seo et al. [129] proposed a model-based switching method to detect ISC. By introducing the ISC resistance to the battery model, the accuracy of the open-circuit voltage (OCV) estimation is improved, and the ISC resistance can be estimated more accurately.



FIGURE 5 - Model-based fault diagnosis for ISC detection. Reproduced with permission from [138] (Copyright 2018 Elsevier).

Another implementation of the model-based method is to combine the information of adjacent cells in a battery pack. Fault diagnosis can be achieved based on the difference of battery states or model parameters between the faulty cell and the normal cell. For example, Feng et al. [138] proposed a model-based ISC fault diagnostic scheme, as shown in Fig. 5. They first calculated the voltages and temperatures of both average and worst cells, and worst cells have the most possibility for an ISC fault. Then, their SOCs and internal ohmic resistances were obtained based on the state estimation and parameter estimation methods, namely EKF and RLS with a forgetting factor. Finally, the ISC fault and its fault level were determined by the deviations of voltage, temperature, SOC, and internal ohmic resistance between batteries. Zhang et al. [141] estimated the resistance of the parallel-connected battery group (PCBG), and identified the capacity fade fault by comparing the PCBG resistance among different PCBGs. Moreover, two fault causes, an inconsistent aging fault, and a loose contact fault, can be distinguished by comparing the PCBG resistances. Ouyang et al. [117] estimated the basic parameters of MDM by the RLS algorithm and then calculated the differential of the voltage and the fluctuation function of the internal resistance. Based on a statistical method, the ISC fault is determined by comparing the estimated and calculated parameters with the threshold. Given that the micro-short circuit (MSC) causes the SOC difference to increase continuously, Gao et al. [36] estimated the SOC difference based on the MDM with an EKF. The extra depleting current is identified, and the short circuit resistance is detected and calculated. Without the need for estimating the SOC of each cell, this method can quantitatively describe the ISC fault with a small computational cost.



FIGURE 6 - A short-circuit fault diagnosis scheme based on the correlation coefficient method. Adapted from [125], [142].

The signal processing method is a typical data-driven method that extracts useful fault features from the battery measurement data directly to detect faults. It does not require the construction of an accurate battery analytical model and is suitable for a wide range of applications. Dubarry et al. [133]-[135] applied incremental capacity analysis (ICA) to identify various contributions to capacity loss, and ICA is more sensitive than traditional charge-discharge curves. The correlation coefficient can be used to determine whether the trends of two voltage curves match with each other. For example, Xia et al. [125] proposed a short-circuit fault diagnosis scheme by using this method, as shown in Fig. 6. The voltage of each cell in the battery pack is readily available, but battery inconsistency makes it difficult to determine battery faults directly by voltage. Therefore, Xia et al. captured the abnormal voltage drop by calculating the correlation coefficient between cell voltages. Then, the short circuit fault was detected by comparing the calculated correlation coefficient with the threshold. According to the mathematical properties of the correlation coefficient algorithm, this method is robust to the inconsistencies in OCV and internal resistance, and the detection process does not require hardware or analytical redundancy. Using the real-time voltage data extracted from the National Service and Management Center of Electric Vehicles (NSMC-EV) in Beijing, Li et al. [142] verified the voltage fault detection of the battery pack based on the interclass correlation coefficient method. Considering that the remaining charging capacity (RCC) of the MSC cell will increase when the battery pack is fully charged each time due to the extra charge depletion, Kong et al. [143] estimated the RCC of each cell based on the uniform charging cell voltage curve (CCVC) hypothesis. According to the difference between the RCCs after two adjacent charges, the leakage current and MSC resistance can be obtained. Based upon a large amount of raw temperature data derived from NSMC-EV in Beijing, Hong et al. [127] applied the Shannon entropy to capture the temperature abnormity of the battery pack. Besides, the abnormity coefficient, including over-temperature and excessive temperature difference, was quantitatively evaluated to predict both the time and location of the temperature faults in battery packs. Wang et al. [128] employed the modified Shannon entropy to analyze the voltage evolution of each cell, and accurately predict both the time and location of the voltage fault in battery packs. Liu et al. [144] regarded all cell voltage values at each time step as an index and implemented the entropy weight method to obtain the objective weight of each index. According to the comprehensive score and the threshold, battery voltage abnormality can be accurately identified.



FIGURE 7 - Electrolyte leakage fault detection based on the random forest classifier. Reproduced with permission from [145] (Copyright 2018 Elsevier).

Another typical data-driven method is machine learning, which learns the underlying laws from a large number of battery training samples. However, it is currently less used in battery fault diagnosis due to the difficulty in obtaining large amounts of battery fault data. For example, Yang et al. [145] proposed a method based on the random forest (RF) classifier to detect electrolyte leakage of ESC cells, as shown in Fig. 7. The leaked cells have a lower discharge capacity and a higher maximum temperature rise. Therefore, these two features were fed into the pre-trained RF model. Firstly, every training subset Si was resampled randomly from the training data set using the Bootstrap method, and then every single decision tree Ci was generated by the corresponding Si. Finally, the output classification results indicate the leakage conditions, which are determined by the voting results of all decision trees. With a large number of offline ESC fault tests, the trained RF classifier can get the correct result rapidly. Zhao et al. [146] combined the 3σ -MSS is utilized to build the criteria of fault-free cell terminal voltages, and a neural network is applied to fit the cell fault distribution in a battery pack. Kim et al. [147], [148] proposed a distance-based outlier detection approach with Z-score standardized pre-processing method for battery fault diagnosis. The estimated capacity and resistance parameters were subjected to cluster analysis for detecting the healthy cells, shorted cells, and aged faulty cells.

Knowledge-based fault diagnosis relies on the understanding of battery mechanisms and the long-term accumulated knowledge and experience. Xiong et al. [149] proposed a rule-based detection method for the over-discharged Li-ion batteries. Based upon the increase of temperature and the decrease of voltage during battery over-discharge, temperature and voltage rules are established respectively, and failure detection and early warning are directly given by a Boolean expression. However, the appropriate fixed or time-varying thresholds in the rules are not easy to be determined in real applications. Muddappa et al. [150] designed an electrochemical model-based observer to generate voltage residual, temperature residual, and SOC residual. Then these residuals along with the temperature change rate, voltage level, and SOC level are all incorporated into the fuzzy rule to detect various fault types including overcharge, over-discharge, and battery aging. Huber et al. [151] proposed a method for classification of battery separator defects using optical inspection, and combined various techniques such as expert knowledge, machine learning, and machine vision in the diagnosis process. This method of integrating multiple diagnostic techniques generally has high precision and robustness, but at the cost of high computational complexity.

DIAGNOSTIC METHODS	REFERENCES	BATTERY FAULTS	FAULT FEATURES	ADVANTAGES	DISADVANTAGES	PRACTICAL LIMITATIONS
RLS	[36], [66], [117], [138], [141]	MSC, ISC, capacity fade fault	Ohmic resistance, capacitance, SOC difference, differential of the voltage.	High precision; Low computational cost.	Affected by model uncertainty and SOC estimation accuracy.	Needs the information of other cells in series; Affected by cell inconsistency and balancing.
PF	[62]	Li plating	Transport rate of Li- ions.	Insensitive to the noise.	Affected by model uncertainty.	High computational cost.
EKF	[59], [60]	Overcharge, overdischarge	Battery model parameters.	Insensitive to the noise.	Complexity of multiple models.	High computational cost.
Luenberger observer	[63]	Thermal faults	Thermal model parameters corresponding to different thermal faults.	Quantitatively assessment of faults.	Sensitive to the noise.	Poor robustness to measurement noise.

TABLE 7 - COMPARISON OF BATTERY FAULT DIAGNOSIS METHODS.

Nonlinear observer-based on Lyapunov	[130]	Thermal Faults	Thermal model parameters.	FDI for three thermal faults.	Affected by model uncertainty.	Needs high model accuracy.
analysis PDE-based observer	[140]	Thermal faults	Thermal model parameters.	High precision.	Complicated PDE model.	High computational cost.
ICA	[133]-[135]	Capacity loss	<i>dQ/dU</i> , capactity.	More sensitive than traditional charge- discharge curves.	Sensitive to the measurement noise.	Affected by battery inconsistency, cycling rate and temperature.
Correlation coefficient	[125], [142]	Short circuit fault, voltage fault	Abnormal voltage drop, correlation coefficient.	Insensitive of the cell inconsistencies; Hardware or analytical redundancy not	Sensitive to the measurement noise.	Affected by the cell balancing.
CCVC transformation	[143]	MSC	RCC.	required. Low computational complexity.	Subject to SOC estimation accuracy.	Difficult to work in real time; Affected by the cell balancing.
Entropy method	[127], [128], [144]	Temperature and voltage fault within battery packs	Entropy of battery temperature and voltage.	Applicable to a wide range, especially abnormal fluctuations of the chaotic system	Computation window has significant effects on the results of entropy.	High computational cost.
RF	[145]	Electrolyte leakage of ESC cells	Discharge capacity and maximum temperature increase.	Good classification ability; Low computational cost.	Needs a large amount of training data.	A large amount of fault data is not easily available.
ANN	[146]	Battery fault	Abnormal voltage.	Accurate; Insensitive of the model uncertainty.	A large amount of training data are required.	A large amount of fault data is not easily available.
Rule-based method	[149]	Overcharge	Increase of temperature and the decrease of voltage.	Easy to implement and understand.	Not easy to determine the appropriate parameters in the rules.	Poor robustness to unknown interferences.
Fuzzy logic	[150]	Overcharge, overdischarge and aged battery	Parameters derived from voltage, temperature and SOC.	Easy to deal with uncertainty in knowledge.	Poor self-learning capability;	High computational cost.
Fusion method of integrating expert knowledge, machine learning, and machine vision	[151]	Battery separator defects	Non-quality related optical effects.	High precision; Good robustness.	High system complexity.	High computational cost.

3.2. Sensor fault diagnosis

3.2.1. Sensor fault mechanisms. In general, the reliability of sensors is affected by manufacturing defects, harsh environment, and working conditions. The sensors in the LIBS discussed in this paper mainly consists of voltage, current, and temperature sensors. Conventional current and voltage sensors used in EV battery systems are Hall effect sensors. Additionally, some advanced technologies, such as constant current source circuit acquisition and isolation amplifier acquisition, are also applied in the monomer voltage acquisition. Thermocouples and resistance temperature detectors are commonly used temperature sensors. For Hall effect sensors, temperature variations can change the magnetic properties of the ferrite core, and there could be some flaws developed in the core, such as corrosion, cracks and core breakage, all of which could result in the bias [152]. Due to mechanical shocks or other

causes that can change the value of Hall voltage, changes in the orientation of the induced magnetic field would lead to a scaling error. For thermocouples, the failure of thermocouple junction, such as corrosion, degradation, and changes in material composition at long-term high temperatures, can lead to bias, scaling, intermittent and/or complete failure [153]. For resistance temperature detectors, overtime exposure in high temperatures, as well as vibration and shock, can change its characteristics, further leading to signal drift [152], [154].

3.2.2. Sensor fault features. Since sensor faults affect the measurement signals directly, the fault features of voltage, current, and temperature sensors are often considered as some forms of bias, drift, scaling, or complete failure signal in sensor measurements [19], [155]. Besides, sensor faults can also be classified into additive and multiplicative faults [37], [45]. In Ref. [155], typical ranges for common sensor faults from literature are summarized, which provides realistic magnitudes to the sensor faults. Voltage measurement is one of the most critical measurements in a battery system due to its high sensitivity to common electrical faults including short circuits, overcharge and overdischarge [156].

3.2.3. Sensor fault diagnosis methods. Table 8 illustrates the sensor fault diagnosis methods used in the battery system, which can be divided into three types: sensor topology-based method, model-based method, and fusion method.

The sensor topology-based method mainly relies on the sensor configuration and redundancy of sensor functionalities, which is easy to implement. Xia et al. [157], [158] proposed a fault-tolerant voltage measurement method for series-connected battery packs by measuring the total voltage of multiple cells instead of measuring the voltage of individual cells. Then, a matrix interpretation of the sensor topology was developed. For this sensor topology, sensor or cell faults can be isolated by locating abnormal signals without additional hardware expense. Kang et al. [159] presented a multi-fault diagnostic scheme that combines voltage measurement topology and correlation coefficient method, in which the correlation coefficient is used to detect fault features. In this sensor topology, each cell and connection resistor is associated with two sensors, which enables isolation of voltage sensor faults, short circuit faults, and connection faults.

The model-based method generates residual by using sensor measurements and a priori information or constraint relationships expressed by the model. After analyzing and evaluating the residuals, the magnitude, type, and location of faults can be determined. Typical battery models for sensor fault diagnosis include the EM [160]-[162], ECM [163]-[167], lumped-parameter thermal models [168], [169], and two-state lumped parameter thermal models [52]. Lombardi et al. [13] tested the electrical relationship between the current sensor and voltage sensor measurements based on Kirchhoff's law to generate residuals, and achieve the FDI of voltage and current sensors according to the battery pack structure and the residual set associated with each sensor. Liu et al. [169] proposed a systematic scheme to apply structural analysis theory to detect and isolate the voltage, current, and temperature sensors faults. Specifically, structural over-determined parts of the system model have been found, and subsequently, fault detectability and isolability analysis have been performed. Then, diagnostic tests are developed by selecting the minimum over-determined set. Finally, the residuals are generated by checking the analytical redundancy relationship in each test. Structural analysis theory [20], [52], [169] can effectively reduce the workload in selecting residual generators. However, this type of analysis is easily affected by noise and model uncertainty.



FIGURE 8 - Flowchart of observer-based sensor fault diagnosis.

Due to the inaccurate initial values, unknown interference, and noise, residuals generated directly through the constraint relationships from a model may carry errors. Observers or filters can reduce the impacts of these factors, and the sensor fault diagnosis based on various observers follows a similar process, as shown in Fig. 8. These methods first estimate the battery states based on the battery model and current, voltage, and temperature sensor measurements. Then, the residuals containing the sensor fault information are generated by comparing the measured outputs and the estimated outputs. Finally, the FDI of sensor faults can be achieved through residual evaluation, and alarms and fault flag should be set. Xu et al. [14] took the current sensor fault as a bias

signal to the system input, and used the proportional integral observer (PIO) to implement fault detection and estimation. Although this method is accurate and easy to implement, improper setting of PIO parameters may cause instability of the diagnostic system. Marcicki et al. [168] provided a scheme based on a modified nonlinear parity equation method to achieve voltage, current and temperature sensor FDI. In the method, a subset of inputs and outputs are hypothesized to be non-faulty. Under this assumption, the residuals are generated by the forward and inverse models of the system, but the minimum detectable fault magnitude is limited by the observer error. By adding the bounded deviation to the sensor measurement, Dey et al. [18] achieved the fault detection, isolation, and estimation of voltage, current, and temperature sensors by using a sliding-mode observer method. Liu et al. [17] presented a model-based diagnostic scheme using EKF to estimate output voltage for detecting current or voltage sensor faults, which is robust to inaccurate initial values and noise. However, the accurate process noise covariance matrix in EKF is not easy to be determined in practice. He et al. [170], [171] achieved the FDI of the current and voltage sensors in the series battery pack based on an adaptive extended Kalman filter (AEKF), which shows better noise robustness because AEKF can adjust the process and measurement noise covariance matrix adaptively.

Strategies combining multiple model-based methods can compensate for the inherent flaws in a single method. For example, Liu et al. [52] constructed two diagnostic tests based on the structural analysis theory. Then, the residuals were generated based on EKF in each diagnostic test. The generated residuals were further evaluated by the statistical cumulative sum test to detect the sensor faults. This fusion scheme reduces the effort required to find the appropriate residual generator and is robust to noise and inaccurate initial values. But it also increases the system complexity and computational cost.

	TABLE 8	- COMPARISON OF	SENSOR FAULT DIAGN	OSIS METHODS.	
DIAGNOSTIC	REFERENCES	ACHIEVEMENTS	ADVANTAGES	DISADVANTAGES	PRACTICAL
METHODS					LIMITATIONS
Fault-tolerant voltage measurement method	[157]-[159]	Fault isolation of sensor or cell.	Able to isolate the sensor fault and cell fault;	High noise level.	Only suitable for series battery packs with interleaved voltage measurements.
Kirchhoff's law	[13]	FDI of voltage, current and temperature sensor.	Simple and low computational cost.	Subject to noise and model uncertainty.	Not suitable for fault estimation.
Structural analysis theory	[20], [52], [169]	FDI of voltage, current and temperature sensor.	Convenient detectability and isolability analysis; Less workload of designing a residual generator.	Highly dependent on redundant information from the system.	Poor robustness to noise and model uncertainty.
PIO	[14]	Fault detection and estimation of current sensor.	Accurate; Easy to implement.	Improper setting of PIO parameters may cause instability.	Needs high model accuracy and proper parameters of PIO.
Nonlinear parity equation method	[168]	FDI of voltage, current and temperature sensor.	Efficient, easy to detect the large fault.	Minimum detectable fault magnitude is limited by the observer error.	Needs high model accuracy; Low sensitivity for fault detection.
Sliding-mode observer	[18]	FDI and fault estimation of voltage, current and temperature sensor.	Good noise robustness.	Sensitive to model uncertainty.	Needs high model accuracy.
EKF	[17]	Fault detection of voltage or current sensor.	Insensitive to noise and inaccurate initial values.	The accurate process noise covariance is not easy to be determined.	Affected by the process noise and model accuracy.
AEKF	[170], [171]	FDI of voltage and current sensor.	Insensitive to noise and inaccurate initial values; Update the noise covariance matrix.	High computational cost.	Needs high model accuracy.
Fusion method integrating EKF and structural analysis theory	[52]	FDI of voltage, current and temperature sensor.	Accurate, low false alarm rate and missed detection rate.	High system complexity.	High computational cost.

3.3. Actuator fault diagnosis

3.3.1. Actuator fault mechanisms. Research on actuator fault diagnosis in the battery system mainly focuses on the battery connection fault and cooling system fault [20], [26]. During EV operations, vibrations may contribute to the loosing or poor electrical connections between batteries [26]. Once the operating and driving voltage becomes too low, the relay and drive motor cannot operate as specified. Fan failures are often caused by electric wire faults or blade damage, which along with motor fault, would severely affect the normal operation of the cooling system.

3.3.2. Actuator fault features. Similar to battery fault features, actuator fault features can also be derived from the system abnormal behavior and equivalent fault parameters. If the battery connection fault occurs, the resistance will increase dramatically and generate significant heat. Moreover, a single high inter-cell contact resistance can cause uneven current flow, resulting in a severe imbalance in the battery pack [172]. Therefore, some features, such as the increased contact resistance, temperature rise, and voltage inconsistency, can be used to characterize the connection fault. In the system model, the battery connection fault can be considered as a gain fault due to the sharp increase in internal resistance caused by a poor connection. For the cooling system fault, it is often considered as an additive fault because it will cause a deviation of the effective heat transfer coefficient in the thermal model.

3.3.3. Actuator fault diagnosis methods. Actuators with different functionalities have various fault mechanisms and features. There is no universal diagnostic method for actuator fault diagnosis. Two typical diagnostic methods are model-based method and signal processing method.

The model-based method can be directly applied to the fault diagnosis of the cooling system. The cooling system, such as the cooling fan or drive motor, is used in the battery system to increase the rate of cooling. The effective heat transfer coefficient, a parameter in the thermal model, varies with the type of convection. Therefore, the cooling system fault could be regarded as the thermal model parameter deviation, which can be detected by typical model-based methods. Liu et al. [20], [169] used the structural analysis theory to implement the cooling system FDI based on the lumped thermal model. Marcicki et al. [168] detected the cooling system fault caused by the failure of the fan motor based on the nonlinear parity equation method.



FIGURE 9 - Battery connection fault diagnosis based on the entropy method. Reproduced with permission from [33] (Copyright 2013 Elsevier).

It is also important to note that the entropy method [173] is an effective tool to describe the degree of disorder of time series, which has a wide range of applications and is suitable for handling actuator faults with abnormal fluctuations, such as battery connection faults. Taheri et al. [174] studied the energy loss caused by the contact resistance of Li-ion battery assemblies for the first time. Zheng et al. [33] proposed an entropy-based connection fault diagnosis scheme, as shown in Fig. 9. Their preliminary analysis identified the two reasons for battery pack power fade, including the internal and contact resistance increase faults. In order

to account for the internal and contact resistance in the resistance calculation, the voltage is usually measured between the ends of the cell and the connecting wire. Then, they established a simplified battery ECM and identified the model parameter containing the contact resistant by the total least squares algorithm. Considering that poor contact conditions between the batteries can make the contact resistance highly unstable, they captured the unstable characteristics of contact resistance by calculating the S hannon entropy of the cell resistance and realized the distinction between cell fault and connection fault. Yao et al. [26] identified the cell connection state by calculating the entropy value of the cell terminal voltage. After a comparative analysis of sample entropy, local Shannon entropy and ensemble Shannon entropy, it is found that the ensemble Shannon entropy can predict the accurate time and the location of a battery connection failure in real-time. Sun et al. [175] used Shannon entropy to process the cell voltage measurements after wavelet transformation, and detected the battery connection fault accurately. Compared with the Shannon entropy iteration method used in Ref. [26], this method simplifies the calculation process and is easier to implement due to the relatively reasonable interval parameters.

3.4. Issues and challenges

Issues and challenges in LIBS fault diagnosis can be divided into two categories: those related to the diagnostic objects and those related to diagnosis or control.

The issues associated with the diagnostic objects are summarized as follows: 1) Many battery fault mechanisms have not been fully understood. For a wide variety of Li-ion batteries, there is no unified understanding of the battery fault mechanisms in the existing literature. 2) Standardized substitute test approaches for battery fault have not been developed. Some destructive methods have poor controllability as well as repeatability, and often instantaneously trigger severe faults, which fails to simulate the incubation phase of a fault. 3) There is no well-established mathematical model for some faults to accurately describe fault behavior, such as models that simulate the growth process of lithium dendrites. 4) The relations between external behaviors and internal mechanisms are not clear. Different conditions could cause the same fault. However, most of the existing research only focuses on a single fault mechanism without considering the coupling between different fault mechanisms.

There are also some diagnostic or control related challenges: 1) The commonly available battery data are voltage, current, and temperature which do not contain any information of the internal electrochemical dynamics in the battery. Extracting the appropriate features to characterize the internal state of the battery still remains a challenge. 2) The internal battery state is difficult to be monitored directly due to the uncertainties with modeling and measurement. 3) The fault data in LIBS are difficult to obtain, which limits the application of data-driven algorithms. 4) The threshold is closely related to false alarm rate, missed detection rate, and time delay of fault detection. The fixed threshold or double-threshold does not meet the requirements in the complex real-world scenarios, and few studies have been conducted to develop the adaptive threshold. 5) It is not easy to detect some minor battery faults at an early stage. However, it could have already caused serious harm to the battery system when it's detected. It is also difficult to correct these unrepairable faults by the battery system itself. 6) Most studies on sensor fault diagnosis and battery fault diagnosis are based on the assumption that other components are trouble-free. Isolating battery fault from sensor fault is still a challenging issue. 7) One of the challenges in the fault diagnosis of the parallel-connected battery pack is that there is no observability and controllability because only one voltage sensor and one current sensor are used. Therefore, for a battery pack that has many parallel-connected battery cells, only the battery pack level faults can be detected. However, the parallel configuration and the traditional sensor placement make it difficult to accurately locate a particular faulty cell in the parallel-connected strings. A better sensor placement design and advanced diagnostic algorithms are needed to solve this problem. 8) The reviewed methods or algorithms provide a wide spectrum of available solutions for the fault diagnosis of battery systems. However, many model-based and data-driven algorithms mentioned in the paper cannot be applied in practical applications currently due to the strict requirements of practical applications. The effectiveness of many model-based and data-driven methods is only tested through simulation studies. Experimental studies are needed to verify their effectiveness. Moreover, the performance of the model-based and data-driven diagnostic algorithms mentioned in this paper is affected by many factors in practical applications, including model accuracy, interference and measurement noise, algorithm robustness, quantity and quality of data, battery inconsistency, sensor topology, and battery pack configuration. When the algorithms are applied to a certain scenario, they should be customized and tuned properly.

4. Future trends of battery system fault diagnostics

Based on the discussion in previous sections, it is clear that the fault diagnosis of the battery system is of multidisciplin ary nature. To develop a robust, reliable, effective battery fault diagnosis system, some important tasks need to be completed for different stages of the fault diagnosis, including the preparation, analysis, and handling, which are summarized in Fig. 10.

4.1. Preparation stage

In the preparation stage, detailed mechanism research, advanced data acquisition, and processing techniques are essential for battery fault diagnosis. Multi-scale mechanism studies from the material, cell, and pack levels help gain an in-depth understanding of the battery system fault. The damage caused by faults could be contained by the fault diagnosis and safety protection at all levels. With the increasing demand for fast charging of EVs, the battery fault mechanism under fast charging conditions should be further investigated. Various side reactions promoted by high-rate charging could contribute to accelerated degradation and TR. Moreover,

it is also important to develop controllable and repeatable fault substitute tests, as well as high-fidelity models for simulating the real faults especially for the ISCs that present the greatest potential threat to battery system safety.

In order to improve the diagnostic performance over the entire lifespan of the battery, the effects of battery aging on diagnostic performance should also be considered. It is important to update and correct the model parameters of the battery, such as cap acity and internal resistance by using advanced techniques including model-based method, machine learning method, and fusion method. Note that the battery ages slowly over time; therefore, the model parameters need to be updated at a long-time-scale or offline.

Data acquisitions using intelligent sensors and integrated chips are also expected, owing to their high accuracy and versatile functionalities. As the sensing technologies evolve, it's also very attractive to use advanced sensors to measure the physical and chemical characteristics within a battery directly. For example, based on the built-in piezoelectric sensors [176], the electrochemical acoustic time of flight analysis can be done to capture the implicit correlation between waveform signal parameters, and battery SOC and SOH. Omega load cell sensors [177] have been applied to measure the cell expansion caused by swelling of the electrode active material during charging. In addition to regular measurements including voltage, current, and temperature, fiber optic sensors [178] are capable of monitoring additional cell parameters, (e.g., strain), and they are also robust to electromagnetic interference. In future research, fast and accurate sensing technologies will continue to be one of the hot topics for a safer battery system.

Feature extraction at the battery pack level is an important task. A number of studies on fault diagnostics of LIBS have been conducted by using the voltage response of battery cells or series-connected battery packs [125], [128], [138], [157]. However, there are few studies on the voltage response of parallel-connected battery packs [141]. It is worth noting that due to the self-balancing mechanism of the parallel structure, a battery fault can also cause transient voltage fluctuations of adjacent parallel cells within the same module.

In the existing literature, the widely used features of thermal-related faults are the temperature rise and fluctuation caused by abnormal heat generations. In fact, thermal-related faults can also affect adjacent cells and lead to uneven temperature distribution in the battery pack. Therefore, useful fault features can be derived by exploring the temperature distribution in the battery pack. Besides, suitable fault features can also be developed by feature transformation or fusion of multiple electrical and thermal features.



FIGURE 10 - Prospects of fault diagnosis technology for Li-ion battery system. Adapted from [179].

4.2. Analysis stage

In the analysis stage, the accurate analyses of the battery system state, including the condition monitoring, fault diagnosis, and fault prognosis, play an important role in the safety of the battery system. The thresholds play a significant role in the trade-off between the sensitivity and robustness for model-based fault diagnosis. Generally, the threshold is affected by a variety of factors, including modeling errors, random disturbances, and system inputs as well as outputs. The accuracy of condition monitoring and fault diagnosis can be improved by developing the adaptive threshold that takes into account battery aging and usage patterns.

The model-based state estimation and parameter estimation methods are still commonly used in the fault diagnosis of LIBS. In comparison with the knowledge-based and data-driven method, the model-based method is more suitable for fault isolation and fault size estimation due to the full utilization of battery system dynamics. For a safer battery system, fault isolation must be included, which identifies and locates a specific fault from battery, sensor, or actuator faults. Besides, due to the significant impacts on the fault severity analysis and subsequent countermeasures, fault estimation should also be considered as one important research topic.

Note that the imbalance of battery capacity, SOC, internal resistance are often ignored in many existing studies. Therefore, future studies should also consider this important factor, especially for battery pack applications. Methods that are robust to battery inconsistency, such as the correlation coefficient method, are expected to be directly used for battery pack fault diagnosis or as a pre-processing technique of machine learning diagnostic methods.

With the advent of the era of big data, data-driven methods are expected to promote the rapid developments of battery system fault diagnosis. However, a single fault diagnosis algorithm has an inherent limitation, and it is often difficult to achieve the desired effect. Therefore, a research trend in fault diagnostics is to fuse multiple fault features and diagnostic algorithms, further improving the overall performance of the diagnostic system.

For some faults with a slow evolution process, such as the spontaneous ISC, early fault diagnostics and prognostics will play an increasingly vital role in ensuring the safety of the battery system. Based on the physics-based model and big data, combining knowledge and data will very likely become an inevitable trend for the next generation of intelligent battery diagnostics.

4.3. Fault handling stage

In the fault handling stage, rapid and efficient actions on the identified battery system faults, such as fault-tolerant control and necessary maintenance, are critical for maintaining the safe operations of a LIBS. Currently, the controller and fault diagnosis subsystems are usually designed separately. It is a potentially promising research topic to develop a fault-tolerant controller that integrates both. Although the prognostics and health management (PHM) for rotating machinery systems have been developed and discussed in Ref. [179], research on the battery system maintenance is still in its infancy. Given the complexity and the poor maintainability of the battery system, it's also an important topic to develop the PHM system.

Currently, the fault diagnosis development for the battery system is shifting from offline to online, from local single-machine control to network-based remote control. The integrated management of monitoring, diagnosis, prognosis, and maintenance over the entire lifespan of the battery would be a future trend in the development of fault diagnostics for LIBS.

5. Conclusions

This paper provides a comprehensive survey on fault mechanisms, fault features, and fault diagnosis of various faults in LIBS, including battery internal faults, sensor faults, and actuator faults. The goal is to provide a comprehensive understanding of the latest technologies and stimulate innovative ideas for LIBS fault diagnosis.

None of the reviewed diagnostic methods is the one-size-fits-all solution for different faults in the battery system. Battery faults have different fault modes, complex fault mechanisms, and coupled relationships between various faults. These faults typicall y result in abnormal changes in estimated battery state and model parameters such as capacity, internal resistance, SOC, and temperature. Therefore, model-based state estimation and parameter estimation have become the most common methods for battery fault diagnosis. Developing high-fidelity battery models can improve the performance of model-based methods. For example, since the battery aging has an important impact on the diagnostic performance, it is important to establish an electro-thermal-aging coupling model and update the model parameters online. There are fewer machine learning-based methods in battery diagnostics because a large amount of fault data in LIBS is not easily available. With the advent of the era of big data, data-driven methods are expected to play an increasingly important role in LIBS fault diagnosis. However, a single fault diagnosis method has inherent limitations, and it is a promising trend to combine multiple fault features and multiple diagnostic methods to improve the accuracy and robustness of LIBS diagnostics further.

Sensor faults and actuator faults are often treated as unknown input signals and model parameter deviations, respectively. Since such faults do not involve the electrochemical information of the battery, a simple ECM is sufficient for the diagnostic requirements in the model-based method. More research work on model-based methods should focus on improving the detection sensitivity of early faults as well as the robustness to model uncertainties, unknown disturbances, and noises. For data-driven methods, the entropy method is particularly suitable for detecting the battery connection faults by capturing the degree of disorder of signals with abnormal fluctuations. Model-based methods capture detailed battery system dynamics and are therefore more suitable for fault isolation and fault size estimation. These methods can be used for fault-tolerant control and subsequent countermeasures.

For fault diagnosis of a battery pack, the differences in voltage, temperature, estimated capacity, SOC, and internal resistance between cells can be taken as effective fault features. Battery fault detection and even short-circuit current estimation can be performed based on the MDM of the battery pack with state estimation and parameter estimation. However, these model-based methods are affected by cell inconsistencies in the battery pack. Therefore, more data-driven methods that are robust to battery inconsistencies, such as the correlation coefficient method, should be developed for LIBS fault diagnosis. Battery fault diagnosis in the parallel-connected battery strings remains a challenge, and a better sensor placement design and advanced diagnostic algorithms are needed to solve this problem.

In conclusion, the fault diagnostics of LIBS is still at its early stage. Battery fault mechanisms under special conditions, such as fast charging, should be further investigated. For some slowly evolving faults, such as the spontaneous ISC, early fault diagnostics and prognostics will play an increasingly important role in ensuring the safety of the battery system. In practice, standardized substitute tests for LIBS faults need to be developed for diagnostic algorithm validation and diagnostic technology development. Thresholds play a significant role in the trade-off between the sensitivity and robustness of fault diagnosis. The accuracy of condition monitoring and fault diagnosis can be improved by developing adaptive thresholds that take into account battery aging and usa ge patterns. In addition, most studies on battery, sensor and actuator fault diagnosis are based on the assumption that other components are fault-free, and therefore multi-faults detection and isolation in battery systems is still a challenging issue. To sum up, advanced LIBS fault diagnosis needs further research on (1) multi-scale mechanism studies at the material, cell, and pack levels; (2) development of battery models and diagnostic methods that can be implemented in practical applications; and (3) multi-fault diagnostics, early fault diagnostics and prognostics, and fault-tolerant control for battery packs.

Acknowledgments

This work was in part supported by National Natural Science Foundation of China (Grant No. 51875054 and No. U1864212) and Chongqing Natural Science Foundation for Distinguished Young Scholars (Grant No. cstc2019jcyjjq0010), Chongqing Science and Technology Bureau, China.

References

- [1] K. Smarsly and K. H. Law, "Decentralized fault detection and isolation in wireless structural health monitoring systems using analytical redundancy," *Adv. Eng. Software*, vol. 73, pp. 1-10, July 2014.
- [2] N. Williard, W. He, C. Hendricks, and M. Pecht, "Lessons learned from the 787 dreamliner issue on lithium-ion battery reliability," *Energies*, vol. 6, no. 9, pp. 4682-4695, 2013.
- [3] S. Abada, G. Marlair, A. Lecocq, M. Petit, V. Sauvant-Moynot, and F. Huet, "Safety focused modeling of lithium-ion batteries: A review," *J. Power Sources*, vol. 306, pp. 178-192, Feb. 2016.
- Z. Kleinman. "Samsung recalls Note 7 flagship over explosive batteries." BBC News. <u>http://www.bbc.com/news/business-37253742</u> (accessed July 8, 2017).
- [5] B. Barnett, "Technologies for detection and intervention of internal short circuits in Li-ion batteries," in *5th Annu. Knowl. Foundation Conf. Battery Safety*, Washington D.C., US, 2014.
- [6] T. M. Bandhauer, S. Garimella, and T. F. Fuller, "A critical review of thermal issues in lithium-ion batteries," *J. Electrochem. Soc.*, vol. 158, no. 3, pp. R1-R25, 2011.
- [7] Q. Wang, P. Ping, X. Zhao, G. Chu, J. Sun, and C. Chen, "Thermal runaway caused fire and explosion of lithium ion battery," *J. Power Sources*, vol. 208, pp. 210-224, June 2012.
- [8] X. Feng, M. Fang, X. He, M. Ouyang, L. Lu, H. Wang, *et al.*, "Thermal runaway features of large format prismatic lithium ion battery using extended volume accelerating rate calorimetry," *J. Power Sources*, vol. 255, pp. 294-301, June 2014.
- [9] X. Feng, J. Sun, M. Ouyang, F. Wang, X. He, L. Lu, *et al.*, "Characterization of penetration induced thermal runaway propagation process within a large format lithium ion battery module," *J. Power Sources*, vol. 275, pp. 261-273, Feb. 2015.
- [10] J. Lamb, C. J. Orendorff, L. A. M. Steele, and S. W. Spangler, "Failure propagation in multi-cell lithium ion batteries," J. Power Sources, vol. 283, pp. 517-523, June 2015.
- [11] X. Feng, L. Lu, M. Ouyang, J. Li, and X. He, "A 3D thermal runaway propagation model for a large format lithium ion battery module," *Energy*, vol. 115, pp. 194-208, Nov. 2016.
- [12] A. Hofmann, N. Uhlmann, C. Ziebert, O. Wiegand, A. Schmidt, and T. Hanemann, "Preventing Li-ion cell explosion during thermal runaway with reduced pressure," *Appl. Therm. Eng.*, vol. 124, pp. 539-544, Sept. 2017.
- [13] W. Lombardi, M. Zarudniev, S. Lesecq, and S. Bacquet, "Sensors fault diagnosis for a BMS," in 2014 European Control Conf. (ECC), Strasbourg, France, 2014, pp. 952-957.
- [14] J. Xu, J. Wang, S. Li, and B. Cao, "A method to simultaneously detect the current sensor fault and estimate the state of energy for batteries in electric vehicles," *Sensors*, vol. 16, no. 8, p. 1328, Aug. 2016.
- [15] X. Hu, H. Jiang, F. Feng, and B. Liu, "An enhanced multi-state estimation hierarchy for advanced lithium-ion battery management," Appl. Energy, vol. 257, p. 114019, Jan. 2020.
- [16] X. Hu, F. Feng, K. Liu, L. Zhang, J. Xie, and B. Liu, "State estimation for advanced battery management: Key challenges and future trends," Renewable Sustainable Energy Rev., vol. 114, p. 109334, Oct. 2019.
- [17] Z. Liu and H. He, "Model-based sensor fault diagnosis of a lithium-ion battery in electric vehicles," *Energies*, vol. 8, no. 7, pp. 6509-6527, 2015.

- [18] S. Dey, S. Mohon, P. Pisu, and B. Ayalew, "Sensor fault detection, isolation, and estimation in lithium-ion batteries," *IEEE Trans. Control Syst. Technol.*, vol. 24, no. 6, pp. 2141-2149, 2016.
- [19] L. Lu, X. Han, J. Li, J. Hua, and M. Ouyang, "A review on the key issues for lithium-ion battery management in electric vehicles," J. Power Sources, vol. 226, pp. 272-288, Mar. 2013.
- [20] Z. Liu, Q. Ahmed, G. Rizzoni, and H. He, "Fault detection and isolation for lithium-ion battery system using structural analysis and sequential residual generation," in *ASME 2014 Dyn. Syst. Control Conf.*, San Antonio, TX, USA, 2014.
- [21] G. Laijiuzhi and D. Xiazhaoer, "Present situation and future prospects of electric vehicles and hybrid electric vehicles," *Automotive Engineering*, vol. 2, pp. 204-209, 2003.
- [22] S. Yonghua, Y. Yuexi, and H. Zechun, "Present status and development trend of batteries for electric vehicles," *Power System Technology*, vol. 35, no. 4, pp. 1-7, 2011.
- [23] G. Liu, M. Ouyang, L. Lu, J. Li, and X. Han, "Analysis of the heat generation of lithium-ion battery during charging and discharging considering different influencing factors," *J. Therm. Anal. Calorim.*, vol. 116, no. 2, pp. 1001-1010, 2014.
- [24] X. N. Feng, J. Sun, M. G. Ouyang, X. M. He, L. U. Lu, X. B. Han, *et al.*, "Characterization of large format lithium ion battery exposed to extremely high temperature," *J. Power Sources*, vol. 272, pp. 457-467, Dec. 2014.
- [25] X. Han, M. Ouyang, L. Lu, and J. Li, "Cycle life of commercial lithium-ion batteries with lithium titanium oxide anodes in electric vehicles," *Energies*, vol. 7, no. 8, pp. 4895-4909, 2014.
- [26] L. Yao, Z. Wang, and J. Ma, "Fault detection of the connection of lithium-ion power batteries based on entropy for electric vehicles," *J. Power Sources*, vol. 293, pp. 548-561, Oct. 2015.
- [27] S. M. M. Alavi, M. F. Samadi, and M. Saif, "Diagnostics in lithium-ion batteries: Challenging issues and recent achievements," in *Integration of Practice-Oriented Knowledge Technology: Trends and Prospectives*, M. Fathi, Ed. Berlin, Heidelberg: Springer Berlin Heidelberg, 2013, pp. 277-291.
- [28] R. J. Patton, "Fault detection and diagnosis in aerospace systems using analytical redundancy," *Computing & Control Engineering Journal*, vol. 2, no. 3, pp. 127-136, 1991.
- [29] Y. Lei, Intelligent fault diagnosis and remaining useful life prediction of rotating machinery. Cambridge: Butterworth-Heinemann, 2016.
- [30] G. Niu, Data-driven technology for engineering systems health management. Singapore: Springer, 2017.
- [31] Z. Liu, X. Yin, Z. Zhang, D. Chen, and W. Chen, "Online rotor mixed fault diagnosis way based on spectrum analysis of instantaneous power in squirrel cage induction motors," *IEEE Trans. Energy Convers.*, vol. 19, no. 3, pp. 485-490, 2004.
- [32] H. Zheng, Z. Li, and X. Chen, "Gear fault diagnosis based on continuous wavelet transform," *Mech. Syst. Sig. Process.*, vol. 16, no. 2-3, pp. 447-457, 2002.
- [33] Y. Zheng, X. Han, L. Lu, J. Li, and M. Ouyang, "Lithium ion battery pack power fade fault identification based on Shannon entropy in electric vehicles," *J. Power Sources*, vol. 223, pp. 136-146, Feb. 2013.
- [34] Q. Zhang, Z. Han, and F. Wen, "A new approach for fault diagnosis in power systems based on rough set theory," in *Int. Conf.* on Advances in Power System Control, Operation and Management, Hong Kong, 1997, pp. 597-602.
- [35] J. Chen and C.-M. Liao, "Dynamic process fault monitoring based on neural network and PCA," *J. Process Control*, vol. 12, no. 2, pp. 277-289, 2002.
- [36] W. Gao, Y. Zheng, M. Ouyang, J. Li, X. Lai, and X. Hu, "Micro-short-circuit diagnosis for series-connected lithium-ion battery packs using mean-difference model," *IEEE Trans. Ind. Electron.*, vol. 66, no. 3, pp. 2132-2142, 2019.
- [37] P. M. Frank, "Fault-diagnosis in dynamic-systems using analytical and knowledge-based redundancy a survey and some new results," *Automatica*, vol. 26, no. 3, pp. 459-474, 1990.
- [38] C. Wu, C. Zhu, Y. Ge, and Y. Zhao, "A review on fault mechanism and diagnosis approach for li-ion batteries," *J. Nanomater.*, vol. 2015, pp. 1-9, Oct. 2015.
- [39] S. M. Rezvanizaniani, Z. Liu, Y. Chen, and J. Lee, "Review and recent advances in battery health monitoring and prognostics technologies for electric vehicle (EV) safety and mobility," *J. Power Sources*, vol. 256, pp. 110-124, June 2014.
- [40] D. Zhou and Y. Hu, "Fault diagnosis techniques for dynamic systems," Acta Autom. Sin., vol. 35, no. 6, pp. 748-758, 2009.
- [41] F. Yang and D. Xiao, "Model and fault inference with the framework of probabilistic SDG," in *9th Int. Conf. Cont. Autom.*, Singapore, Singapore, 2006, pp. 1-6.
- [42] M. E. Paté-Cornell, "Fault trees vs. event trees in reliability analysis," Risk Anal., vol. 4, no. 3, pp. 177-186, 1984.
- [43] M. Held and R. Brönnimann, "Safe cell, safe battery? Battery fire investigation using FMEA, FTA and practical experiments," *Microelectron. Reliab.*, vol. 64, pp. 705-710, Sept. 2016.
- [44] F. Filippetti, M. Martelli, G. Franceschini, and C. Tassoni, "Development of expert system knowledge base to on-line diagnosis of rotor electrical faults of induction motors," in *1992 IEEE Ind. Applic. Soc. Ann. Meeting*, Houston, TX, USA, USA, 1992, pp. 92-99.
- [45] R. Isermann, "Model-based fault-detection and diagnosis status and applications," *Annual Reviews in Control*, vol. 29, no. 1, pp. 71-85, 2005.
- [46] I. Hwang, S. Kim, Y. Kim, and C. E. Seah, "A survey of fault detection, isolation, and reconfiguration methods," *IEEE Trans. Control Syst. Technol.*, vol. 18, no. 3, pp. 636-653, 2010.
- [47] X. Hu, W. Liu, X. Lin, and Y. Xie, "A comparative study of control-oriented thermal models for cylindrical li-ion batteries," IEEE Trans. Transp. Electrif., pp. 1, Nov. 2019. doi: 10.1109/TTE.2019.29536066.
- [48] F. Bagheri, H. Khaloozaded, and K. Abbaszadeh, "Stator fault detection in induction machines by parameter estimation, using adaptive kalman filter," in 2007 *Mediterranean Conf. Control Autom.*, Athens, Greece, 2007, pp. 1-6.
- [49] X.-Q. Liu, H.-Y. Zhang, J. Liu, and J. Yang, "Fault detection and diagnosis of permanent-magnet DC motor based on parameter estimation and neural network," *IEEE Trans. Ind. Electron.*, vol. 47, no. 5, pp. 1021-1030, 2000.
- [50] H. M. Odendaal and T. Jones, "Actuator fault detection and isolation: An optimised parity space approach," *Control Eng. Pract.*, vol. 26, pp. 222-232, May 2014.
- [51] M. Staroswiecki and G. Comtet-Varga, "Analytical redundancy relations for fault detection and isolation in algebraic dynamic systems," *Automatica*, vol. 37, no. 5, pp. 687-699, 2001.

- [52] Z. Liu, Q. Ahmed, J. Zhang, G. Rizzoni, and H. He, "Structural analysis based sensors fault detection and isolation of cylindrical lithium-ion batteries in automotive applications," *Control Eng. Pract.*, vol. 52, pp. 46-58, July 2016.
- [53] M. Krysander and E. Frisk, "Sensor placement for fault diagnosis," *IEEE Trans. Syst. Man Cybern. Part A Syst. Humans*, vol. 38, no. 6, pp. 1398-1410, 2008.
- [54] C. Svärd and M. Nyberg, "Automated design of an FDI-system for the wind turbine benchmark," *IFAC Proceedings Volumes*, vol. 44, no. 1, pp. 8307-8315, 2011.
- [55] C. Svärd, M. Nyberg, E. Frisk, and M. Krysander, "Automotive engine FDI by application of an automated model-based and data-driven design methodology," *Control Eng. Pract.*, vol. 21, no. 4, pp. 455-472, 2013.
- [56] M. Krysander, J. Aslund, and M. Nyberg, "An efficient algorithm for finding minimal overconstrained subsystems for modelbased diagnosis," *IEEE Trans. Syst. Man Cybern. Part A Syst. Humans*, vol. 38, no. 1, pp. 197-206, 2008.
- [57] C. Svärd, M. Nyberg, and E. Frisk, "A greedy approach for selection of residual generators," in 22nd Int. Workshop on *Principles of Diagnosis*, Murnau, Germany, 2011.
- [58] C. Svard and M. Nyberg, "Residual generators for fault diagnosis using computation sequences with mixed causality applied to automotive systems," *IEEE Trans. Syst. Man Cybern. Part A Syst. Humans,* vol. 40, no. 6, pp. 1310-1328, 2010.
- [59] A. Singh, A. Izadian, and S. Anwar, "Fault diagnosis of Li-Ion batteries using multiple-model adaptive estimation," in *IECON 2013-39th Ann. Conf. of the IEEE Ind. Elec.*, Vienna, Austria, 2013, pp. 3524-3529.
- [60] A. Sidhu, A. Izadian, and S. Anwar, "Adaptive nonlinear model-based fault diagnosis of li-ion batteries," *IEEE Trans. Ind. Electron.*, vol. 62, no. 2, pp. 1002-1011, 2015.
- [61] M. Das, S. Sadhu, and T. K. Ghoshal, "Fault detection and isolation using an adaptive unscented Kalman filter," *IFAC Proceedings Volumes*, vol. 47, no. 1, pp. 326-332, 2014.
- [62] S. M. M. Alavi, M. F. Samadi, and M. Saif, "Plating mechanism detection in lithium-ion batteries, by using a particle-filtering based estimation technique," in *2013 Amer. Contr. Conf.*, Washington, DC, USA, 2013, pp. 4356-4361.
- [63] S. Dey, Z. A. Biron, S. Tatipamula, N. Das, S. Mohon, B. Ayalew, *et al.*, "On-board thermal fault diagnosis of lithium-ion batteries for hybrid electric vehicle application," *IFAC-PapersOnLine*, vol. 48, no. 15, pp. 389-394, 2015.
- [64] X. Lin, H. E. Perez, J. B. Siegel, A. G. Stefanopoulou, Y. Li, R. D. Anderson, *et al.*, "Online parameterization of lumped thermal dynamics in cylindrical lithium ion batteries for core temperature estimation and health monitoring," *IEEE Trans. Control Syst. Technol.*, vol. 21, no. 5, pp. 1745-1755, 2013.
- [65] S. Park and D. M. Himmelblau, "Fault detection and diagnosis via parameter estimation in lumped dynamic systems," *Ind. Eng. Chem. Process Des. Dev.*, vol. 22, no. 3, pp. 482-487, 1983.
- [66] X. Feng, C. Weng, M. Ouyang, and J. Sun, "Online internal short circuit detection for a large format lithium ion battery," *Appl. Energy*, vol. 161, pp. 168-180, Jan. 2016.
- [67] R. Isermann and P. Ballé, "Trends in the application of model-based fault detection and diagnosis of technical processes," *Control Eng. Pract.*, vol. 5, no. 5, pp. 709-719, 1997.
- [68] R. Isermann, "Process fault detection based on modeling and estimation methods—A survey," *automatica*, vol. 20, no. 4, pp. 387-404, 1984.
- [69] M. Kinnaert, "Design of redundancy relations for failure detection and isolation by constrained optimization," *Int. J. Control*, vol. 63, no. 3, pp. 609-622, 1996.
- [70] V. Venkatasubramanian, R. Rengaswamy, and S. N. Kavuri, "A review of process fault detection and diagnosis. Part II: Quanlitative models and search strategies. Comput. Chem. Eng. 27(3), 313-326," *Comput. Chem. Eng.*, vol. 27, no. 3, pp. 313-326, 2003.
- [71] B. Samanta and K. Al-Balushi, "Artificial neural network based fault diagnostics of rolling element bearings using timedomain features," *Mech. Syst. Sig. Process.*, vol. 17, no. 2, pp. 317-328, 2003.
- [72] A. Widodo and B.-S. Yang, "Support vector machine in machine condition monitoring and fault diagnosis," *Mech. Syst. Sig. Process.*, vol. 21, no. 6, pp. 2560-2574, 2007.
- [73] L. Jack and A. Nandi, "Support vector machines for detection and characterization of rolling element bearing faults," *Proceedings of the Institution of Mechanical Engineers, Part C: Journal of Mechanical Engineering Science*, vol. 215, no. 9, pp. 1065-1074, 2001.
- [74] D. Zhou and D. X, "Theory and applications of fault tolerant control," *Acta Autom. Sin.*, vol. 26, no. 6, pp. 788-797, 2000.
- [75] R. J. Patton, "Fault-tolerant control: The 1997 situation," *IFAC Proceedings Volumes*, vol. 30, no. 18, pp. 1029-1051, 1997.
- [76] L. Fu, "Research on fault diagnosis and fault-tolerant control of multi-agent system," M.S. thesis, Department of Electrical and Electronic Engineering, Changchun University of Technology, Changchu, China, 2017.
- [77] C. Hu, R. He, R. Wang, and J. Yu, "Fault prediction and fault-tolerant of lithium-ion batteries temperature failure for electric vehicle," in *3rd Int. Conf. Digit. Manuf. & Autom.*, GuiLin, China, 2012, pp. 410-413.
- [78] G. Berdichevsky, K. Kelty, J. Straubel, and E. Toomre, "The tesla roadster battery system," *Tesla Motors*, vol. 1, no. 5, pp. 1-5, Aug. 2007.
- [79] Y. Du, "Research on decentralized control and active fault tolerant control for reconfigurable manipulator," PhD Thesis, Department of Communication Engineering, Jilin University, Changchun, China, 2014.
- [80] M. Blanke, M. Kinnaert, J. Lunze, M. Staroswiecki, and J. Schröder, *Diagnosis and fault-tolerant control*, Third ed. vol. 2. Berlin, Heidelberg: Springer, 2016.
- [81] S. Hu, D. Xu, and Y. Liu, "Evaluation on fault diagnosis methods of dynamic system," *Journal of North China Electric Power University*, vol. 30, no. 6, pp. 20-24, 2003.
- [82] R. J. Patton, P. M. Frank, and R. N. Clarke, *Fault diagnosis in dynamic systems: theory and application*. Upper Saddle River, NJ, USA: Prentice-Hall, Inc., 1989.
- [83] S. Dash and V. Venkatasubramanian, "Challenges in the industrial applications of fault diagnostic systems," *Comput. Chem. Eng.*, vol. 24, no. 2-7, pp. 785-791, 2000.
- [84] D. Aurbach, E. Zinigrad, Y. Cohen, and H. Teller, "A short review of failure mechanisms of lithium metal and lithiated graphite anodes in liquid electrolyte solutions," *Solid State Ionics*, vol. 148, no. 3-4, pp. 405-416, Jun 2002.

- [85] C. Hendricks, N. Williard, S. Mathew, and M. Pecht, "A failure modes, mechanisms, and effects analysis (FMMEA) of lithiumion batteries," J. Power Sources, vol. 297, pp. 113-120, Nov. 2015.
- D. Lyu, B. Ren, and S. Li, "Failure modes and mechanisms for rechargeable Lithium-based batteries: a state-of-the-art review," [86] Acta Mech., vol. 230, no. 3, pp. 701-727, 2018.
- [87] P. G. Balakrishnan, R. Ramesh, and T. Prem Kumar, "Safety mechanisms in lithium-ion batteries," J. Power Sources, vol. 155, no. 2, pp. 401-414, 2006.
- [88] W. Chao, Z. Chunbo, S. Jinlei, and J. Jianhu, "Fault mechanism study on Li-ion battery at over-discharge and its diagnosis approach," IET Electr. Syst. Transp., vol. 7, no. 1, pp. 48-54, 2017.
- [89] P. Arora, R. E. White, and M. Doyle, "Capacity fade mechanisms and side reactions in lithium-ion batteries," J. Electrochem. Soc., vol. 145, no. 10, pp. 3647-3667, 1998.
- [90] H. Maleki and J. N. Howard, "Effects of overdischarge on performance and thermal stability of a Li-ion cell," J. Power Sources, vol. 160, no. 2, pp. 1395-1402, 2006.
- [91] K. W. Nam, S. M. Bak, E. Hu, X. Yu, Y. Zhou, X. Wang, et al., "Combining in situ synchrotron X-ray diffraction and absorption techniques with transmission electron microscopy to study the origin of thermal instability in overcharged cathode materials for lithium-ion batteries," Adv. Funct. Mater., vol. 23, no. 8, pp. 1047-1063, 2013.
- V. A. Agubra and J. W. Fergus, "The formation and stability of the solid electrolyte interface on the graphite anode," J. Power [92] Sources, vol. 268, pp. 153-162, Dec. 2014.
- [93] P. Ramadass, B. Haran, P. M. Gomadam, R. White, and B. N. Popov, "Development of first principles capacity fade model for li-ion cells," J. Electrochem. Soc., vol. 151, no. 2, p. A196, 2004.
- [94] H. Li, J. Gao, and S. Zhang, "Effect of overdischarge on swelling and recharge performance of lithium ion cells," Chin. J. Chem., vol. 26, no. 9, pp. 1585-1588, 2008.
- [95] L. Zhang, Y. Ma, X. Cheng, C. Du, T. Guan, Y. Cui, et al., "Capacity fading mechanism during long-term cycling of overdischarged LiCoO2/mesocarbon microbeads battery," J. Power Sources, vol. 293, pp. 1006-1015, Oct. 2015.
- [96] S. Erol, M. E. Orazem, and R. P. Muller, "Influence of overcharge and over-discharge on the impedance response of LiCoO2|C batteries," J. Power Sources, vol. 270, pp. 92-100, Dec. 2014.
- J. Wen, Y. Yu, and C. Chen, "A review on lithium-ion batteries safety issues: Existing problems and possible solutions," [97] Materials Express, vol. 2, no. 3, pp. 197-212, 2012.
- X. Hu, Y. Zheng, D. A. Howey, H. Perez, A. Foley, and M. Pecht, "Battery warm-up methodologies at subzero temperatures [98] for automotive applications: Recent advances and perspectives," Prog. Energy Combust. Sci., vol. 77, p. 100806, Mar. 2020.
- [99] X. Feng, M. Ouyang, X. Liu, L. Lu, Y. Xia, and X. He, "Thermal runaway mechanism of lithium ion battery for electric vehicles: A review," Energy Storage Mater., vol. 10, pp. 246-267, Jan. 2018.
- [100] H. Wang, S. Simunovic, H. Maleki, J. N. Howard, and J. A. Hallmark, "Internal configuration of prismatic lithium-ion cells at the onset of mechanically induced short circuit," *J. Power Sources*, vol. 306, pp. 424-430, Feb. 2016. S. H. Chung, T. Tancogne-Dejean, J. Zhu, H. Luo, and T. Wierzbicki, "Failure in lithium-ion batteries under transverse
- [101] indentation loading," J. Power Sources, vol. 389, pp. 148-159, June 2018.
- [102] B. Mao, H. Chen, Z. Cui, T. Wu, and Q. Wang, "Failure mechanism of the lithium ion battery during nail penetration," Int. J. Heat Mass Transfer, vol. 122, pp. 1103-1115, July 2018.
- [103] D. H. Doughty and E. P. Roth, "A general discussion of Li ion battery safety," The Electrochemical Society Interface, vol. 21, no. 2, pp. 37-44, 2012.
- [104] R. Zhao, J. Liu, and J. Gu, "Simulation and experimental study on lithium ion battery short circuit," Appl. Energy, vol. 173, pp. 29-39, July 2016.
- [105] W. Fang, P. Ramadass, and Z. Zhang, "Study of internal short in a Li-ion cell-II. Numerical investigation using a 3D electrochemical-thermal model," J. Power Sources, vol. 248, pp. 1090-1098, Feb. 2014.
- H. Maleki and J. N. Howard, "Internal short circuit in Li-ion cells," J. Power Sources, vol. 191, no. 2, pp. 568-574, 2009. [106]
- [107] X. Feng, "Thermal runaway initiation and propagation of lithium-ion traction battery for electric vehicle: Test, modeling and prevention," Ph.D. dissertation, Department of Automotive Engineering, Tsinghua University, Beijing, China, 2016.
- [108] W. Cai, H. Wang, H. Maleki, J. Howard, and E. Lara-Curzio, "Experimental simulation of internal short circuit in Li-ion and Li-ion-polymer cells," J. Power Sources, vol. 196, no. 18, pp. 7779-7783, 2011.
- [109] E. Sahraei, E. Bosco, B. Dixon, and B. Lai, "Microscale failure mechanisms leading to internal short circuit in Li-ion batteries under complex loading scenarios," J. Power Sources, vol. 319, pp. 56-65, July 2016.
- [110] Q. Wang, P. Huang, P. Ping, Y. Du, K. Li, and J. Sun, "Combustion behavior of lithium iron phosphate battery induced by external heat radiation," J. Loss Prev. Process Ind., vol. 49, pp. 961-969, Sept. 2017.
- J. Ye, H. Chen, Q. Wang, P. Huang, J. Sun, and S. Lo, "Thermal behavior and failure mechanism of lithium ion cells during [111] overcharge under adiabatic conditions," Appl. Energy, vol. 182, pp. 464-474, Nov. 2016.
- [112] K. Kitoh and H. Nemoto, "100 Wh Large size Li-ion batteries and safety tests," J. Power Sources, vol. 81, pp. 887-890, Sept. 1999.
- D. P. Abraham, J. Liu, C. H. Chen, Y. E. Hyung, M. Stoll, N. Elsen, et al., "Diagnosis of power fade mechanisms in high-[113] power lithium-ion cells," J. Power Sources, vol. 119-121, pp. 511-516, June 2003.
- [114] N. S. Spinner, C. R. Field, M. H. Hammond, B. A. Williams, K. M. Myers, A. L. Lubrano, et al., "Physical and chemical analysis of lithium-ion battery cell-to-cell failure events inside custom fire chamber," J. Power Sources, vol. 279, pp. 713-721, Apr. 2015.
- D. P. Abraham, E. P. Roth, R. Kostecki, K. McCarthy, S. MacLaren, and D. H. Doughty, "Diagnostic examination of thermally [115] abused high-power lithium-ion cells," J. Power Sources, vol. 161, no. 1, pp. 648-657, Oct. 2006.
- [116] P. Ramadass, W. Fang, and Z. Zhang, "Study of internal short in a Li-ion cell I. Test method development using infra-red imaging technique," J. Power Sources, vol. 248, pp. 769-776, Feb. 2014.
- M. Ouyang, M. Zhang, X. Feng, L. Lu, J. Li, X. He, et al., "Internal short circuit detection for battery pack using equivalent [117] parameter and consistency method," J. Power Sources, vol. 294, pp. 272-283, Oct. 2015.

- [118] C. Park and A. K. Jaura, "Dynamic thermal model of li-ion battery for predictive behavior in hybrid and fuel cell vehicles," SAE Trans., vol. 112, no. 3, pp. 1835-1842, 2003.
- [119] S. Santhanagopalan, P. Ramadass, and J. Zhang, "Analysis of internal short-circuit in a lithium ion cell," J. Power Sources, vol. 194, no. 1, pp. 550-557, 2009.
- [120] S. Dey and B. Ayalew, "A diagnostic scheme for detection, isolation and estimation of electrochemical faults in lithium-ion cells," in *ASME 2015 Dyn. Syst. Control Conf.*, Columbus, Ohio, USA, 2015, pp. V001T13A001-V001T13A001.
- [121] G.-H. Kim, K. Smith, J. Ireland, and A. Pesaran, "Fail-safe design for large capacity lithium-ion battery systems," J. Power Sources, vol. 210, pp. 243-253, July 2012.
- [122] C. Zhang, S. Santhanagopalan, M. A. Sprague, and A. A. Pesaran, "A representative-sandwich model for simultaneously coupled mechanical-electrical-thermal simulation of a lithium-ion cell under quasi-static indentation tests," *J. Power Sources*, vol. 298, pp. 309-321, Dec. 2015.
- [123] C. Zhang, S. Santhanagopalan, M. A. Sprague, and A. A. Pesaran, "Coupled mechanical-electrical-thermal modeling for shortcircuit prediction in a lithium-ion cell under mechanical abuse," *J. Power Sources*, vol. 290, pp. 102-113, Sept. 2015.
- [124] X. Feng, X. He, L. Lu, and M. Ouyang, "Analysis on the fault features for internal short circuit detection using an electrochemical-thermal coupled model," *J. Electrochem. Soc.*, vol. 165, no. 2, pp. A155-A167, 2018.
- [125] B. Xia, Y. Shang, T. Nguyen, and C. Mi, "A correlation based fault detection method for short circuits in battery packs," *J. Power Sources*, vol. 337, pp. 1-10, Jan. 2017.
- [126] E. Sahraei, J. Campbell, and T. Wierzbicki, "Modeling and short circuit detection of 18650 Li-ion cells under mechanical abuse conditions," *J. Power Sources*, vol. 220, pp. 360-372, Dec. 2012.
- [127] J. Hong, Z. Wang, and P. Liu, "Big-data-based thermal runaway prognosis of battery systems for electric vehicles," *Energies*, vol. 10, no. 7, p. 919, 2017.
- [128] Z. Wang, J. Hong, P. Liu, and L. Zhang, "Voltage fault diagnosis and prognosis of battery systems based on entropy and Z score for electric vehicles," *Appl. Energy*, vol. 196, pp. 289-302, June 2017.
- [129] M. Seo, T. Goh, M. Park, G. Koo, and S. Kim, "Detection of internal short circuit in lithium ion battery using model-based switching model method," *Energies*, vol. 10, no. 1, p. 76, 2017.
- [130] S. Dey, Z. A. Biron, S. Tatipamula, N. Das, S. Mohon, B. Ayalew, *et al.*, "Model-based real-time thermal fault diagnosis of Lithium-ion batteries," *Control Eng. Pract.*, vol. 56, pp. 37-48, Nov. 2016.
- [131] W. Liu, "Research and implemention of failure diagnosis expert system for battery pack," M.S. thesis, Department of Software, Hunan University, Changsha, China, 2005.
- [132] J. Wu, "Design of remote monitoring system and study on fault diagnosis method for electric vehicle," M.S. thesis, Department of Automotive Engineering, Jilin University, Changchun, China, 2011.
- [133] M. Dubarry, C. Truchot, and B. Y. Liaw, "Cell degradation in commercial LiFePO4 cells with high-power and high-energy designs," J. Power Sources, vol. 258, pp. 408-419, July 2014.
- [134] M. Dubarry and B. Y. Liaw, "Identify capacity fading mechanism in a commercial LiFePO4 cell," J. Power Sources, vol. 194, no. 1, pp. 541-549, 2009.
- [135] I. Bloom, J. P. Christophersen, D. P. Abraham, and K. L. Gering, "Differential voltage analyses of high-power lithium-ion cells," J. Power Sources, vol. 1, no. 157, pp. 537-542, 2006.
- [136] J. Vetter, P. Novák, M. R. Wagner, C. Veit, K. C. Möller, J. O. Besenhard, et al., "Ageing mechanisms in lithium-ion batteries," J. Power Sources, vol. 147, no. 1-2, pp. 269-281, 2005.
- [137] X. Zhou, T. Ersal, J. L. Stein, and D. S. Bernstein, "Battery state of health monitoring by side reaction current density estimation via retrospective-cost subsystem identification," in ASME 2014 Dyn. Syst. Control Conf., San Antonio, TX, USA, 2014, pp. V001T19A002-V001T19A002.
- [138] X. Feng, Y. Pan, X. He, L. Wang, and M. Ouyang, "Detecting the internal short circuit in large-format lithium-ion battery using model-based fault-diagnosis algorithm," *J. Storage Mater.*, vol. 18, pp. 26-39, Aug. 2018.
- [139] W. Chen, W.-T. Chen, M. Saif, M.-F. Li, and H. Wu, "Simultaneous fault isolation and estimation of lithium-ion batteries via synthesized design of Luenberger and learning observers," *IEEE Trans. Control Syst. Technol.*, vol. 22, no. 1, pp. 290-298, 2014.
- [140] S. Dey, H. E. Perez, and S. J. Moura, "Model-based battery thermal fault diagnostics: Algorithms, analysis, and experiments," *IEEE Trans. Control Syst. Technol.*, vol. 27, no. 2, pp. 576-587, 2019.
- [141] H. Zhang, L. Pei, J. Sun, K. Song, R. Lu, Y. Zhao, *et al.*, "Online diagnosis for the capacity fade fault of a parallel-connected lithium ion battery group," *Energies*, vol. 9, no. 5, p. 387, 2016.
- [142] X. Li and Z. Wang, "A novel fault diagnosis method for lithium-Ion battery packs of electric vehicles," *Measurement*, vol. 116, pp. 402-411, Feb. 2018.
- [143] X. Kong, Y. Zheng, M. Ouyang, L. Lu, J. Li, and Z. Zhang, "Fault diagnosis and quantitative analysis of micro-short circuits for lithium-ion batteries in battery packs," J. Power Sources, vol. 395, pp. 358-368, Aug. 2018.
- [144] P. Liu, Z. Sun, Z. Wang, and J. Zhang, "Entropy-based voltage fault diagnosis of battery systems for electric vehicles," *Energies*, vol. 11, no. 1, p. 136, 2018.
- [145] R. Yang, R. Xiong, H. He, and Z. Chen, "A fractional-order model-based battery external short circuit fault diagnosis approach for all-climate electric vehicles application," J. Cleaner Prod., vol. 187, pp. 950-959, June 2018.
- [146] Y. Zhao, P. Liu, Z. Wang, L. Zhang, and J. Hong, "Fault and defect diagnosis of battery for electric vehicles based on big data analysis methods," *Appl. Energy*, vol. 207, pp. 354-362, Dec. 2017.
- [147] T. Kim, A. Adhikaree, R. Pandey, D. Kang, M. Kim, C. Oh, *et al.*, "Outlier mining-based fault diagnosis for multicell lithiumion batteries using a low-priced microcontroller," in 2018 IEEE Appl. Power Elec. Conf., San Antonio, TX, USA, 2018, pp. 3365-3369.
- [148] T. Kim, D. Makwana, A. Adhikaree, J. Vagdoda, and Y. Lee, "Cloud-based battery condition monitoring and fault diagnosis platform for large-scale lithium-ion battery energy storage systems," *Energies*, vol. 11, no. 1, p. 125, 2018.

- [149] J. Xiong, H. Banvait, L. Li, Y. Chen, J. Xie, Y. Liu, et al., "Failure detection for over-discharged Li-ion batteries," in 2012 IEEE Int. Electric Vehicle Conf., Greenville, SC, USA, 2012, pp. 1-5.
- [150] V. K. Muddappa and S. Anwar, "Electrochemical model based fault diagnosis of li-ion battery using fuzzy logic," in ASME 2014 Int. Mechanical Engineering Congr. and Expo., Montreal, Quebec, Canada, 2014, pp. V04BT04A048-V04BT04A048.
- [151] J. Huber, C. Tammer, S. Krotil, S. Waidmann, X. Hao, C. Seidel, *et al.*, "Method for classification of battery separator defects using optical inspection," *Procedia CIRP*, vol. 57, pp. 585-590, Nov. 2016.
- [152] J. G. Webster, The measurement, instrumentation and sensors handbook. Boca Raton: CRC press, 1998.
- [153] T. Hamada and Y. Suyama, "EMF drift and inhomogeneity of type K thermocouples," in SICE 2004 ann. Conf., Sapporo, Japan, 2004, pp. 989-992.
- [154] R. H. Bishop, *The mechatronics handbook 2 volume set*. Boca Raton: CRC press, 2002.
- [155] E. Balaban, A. Saxena, P. Bansal, K. F. Goebel, and S. Curran, "Modeling, detection, and disambiguation of sensor faults for aerospace applications," *IEEE Sens. J.*, vol. 9, no. 12, pp. 1907-1917, 2009.
- [156] B. Xia, Z. Chen, C. Mi, and B. Robert, "External short circuit fault diagnosis for lithium-ion batteries," in 2014 IEEE Transp. Elect. Conf. Expo., Dearborn, MI, USA, 2014, pp. 1-7.
- [157] B. Xia and C. Mi, "A fault-tolerant voltage measurement method for series connected battery packs," J. Power Sources, vol. 308, pp. 83-96, Mar. 2016.
- [158] B. Xia, T. Nguyen, J. Yang, and C. Mi, "The improved interleaved voltage measurement method for series connected battery packs," *J. Power Sources*, vol. 334, pp. 12-22, Dec. 2016.
- [159] Y. Kang, B. Duan, Z. Zhou, Y. Shang, and C. Zhang, "A multi-fault diagnostic method based on an interleaved voltage measurement topology for series connected battery packs," J. Power Sources, vol. 417, pp. 132-144, Mar. 2019.
- [160] T. F. Fuller, M. Doyle, and J. Newman, "Simulation and optimization of the dual lithium ion insertion cell," J. Electrochem. Soc., vol. 141, no. 1, pp. 1-10, 1994.
- [161] P. M. Gomadam, J. W. Weidner, R. A. Dougal, and R. E. White, "Mathematical modeling of lithium-ion and nickel battery systems," J. Power Sources, vol. 110, no. 2, pp. 267-284, 2002.
- [162] K. A. Smith, C. D. Rahn, and C. Wang, "Model-based electrochemical estimation of lithium-ion batteries," in 2008 IEEE Int. Conf. on Contr., San Antonio, TX, USA, 2008, pp. 714-719.
- [163] F. Huet, "A review of impedance measurements for determination of the state-of-charge or state-of-health of secondary batteries," J. Power Sources, vol. 70, no. 1, pp. 59-69, 1998.
- [164] S. Piller, M. Perrin, and A. Jossen, "Methods for state-of-charge determination and their applications," *J. Power Sources*, vol. 96, no. 1, pp. 113-120, 2001.
- [165] J. Lee, O. Nam, and B. H. Cho, "Li-ion battery SOC estimation method based on the reduced order extended Kalman filtering," J. Power Sources, vol. 174, no. 1, pp. 9-15, 2007.
- [166] X. Tang, X. Mao, J. Lin, and B. Koch, "Li-ion battery parameter estimation for state of charge," in 2011 Amer. Contr. Conf., San Francisco, CA, USA, 2011, pp. 941-946.
- [167] X. Hu, S. Li, and H. Peng, "A comparative study of equivalent circuit models for Li-ion batteries," J. Power Sources, vol. 198, pp. 359-367, Jan. 2012.
- [168] J. Marcicki, S. Onori, and G. Rizzoni, "Nonlinear fault detection and isolation for a lithium-ion battery management system," in ASME 2010 Dyn. Syst. Control Conf., Cambridge, Massachusetts, USA, 2010, pp. 607-614.
- [169] Z. Liu, H. He, Q. Ahmed, and G. Rizzoni, "Structural analysis based fault detection and isolation applied for a lithium-ion battery pack," *IFAC-PapersOnLine*, vol. 48, no. 21, pp. 1465-1470, 2015.
- [170] H. He, Z. Liu, and Y. Hua, "Adaptive extended kalman filter based fault detection and isolation for a lithium-ion battery pack," *Energy Procedia*, vol. 75, pp. 1950-1955, Aug. 2015.
- [171] Z. Liu and H. He, "Sensor fault detection and isolation for a lithium-ion battery pack in electric vehicles using adaptive extended Kalman filter," *Appl. Energy*, vol. 185, pp. 2033-2044, Jan. 2017.
- [172] G. J. Offer, V. Yufit, D. A. Howey, B. Wu, and N. P. Brandon, "Module design and fault diagnosis in electric vehicle batteries," *J. Power Sources*, vol. 206, pp. 383-392, May 2012.
- [173] J. S. Richman and J. R. Moorman, "Physiological time-series analysis using approximate entropy and sample entropy," *American Journal of Physiology-Heart and Circulatory Physiology*, vol. 278, no. 6, pp. H2039-H2049, 2000.
- [174] P. Taheri, S. Hsieh, and M. Bahrami, "Investigating electrical contact resistance losses in lithium-ion battery assemblies for hybrid and electric vehicles," *J. Power Sources*, vol. 196, no. 15, pp. 6525-6533, 2011.
- [175] Z. Sun, P. Liu, and Z. Wang, "Real-time fault diagnosis method of battery system based on Shannon entropy," *Energy Procedia*, vol. 105, pp. 2354-2359, May 2017.
- [176] P. Ladpli, F. Kopsaftopoulos, and F.-K. Chang, "Estimating state of charge and health of lithium-ion batteries with guided waves using built-in piezoelectric sensors/actuators," *J. Power Sources*, vol. 384, pp. 342-354, Apr. 2018.
- [177] Y. Kim, N. A. Samad, K.-Y. Oh, J. B. Siegel, B. I. Epureanu, and A. G. Stefanopoulou, "Estimating state-of-charge imbalance of batteries using force measurements," in 2016 Amer. Confr. Conf., Boston, MA, USA, 2016, pp. 1500-1505.
- [178] L. W. Sommer, A. Raghavan, P. Kiesel, B. Saha, T. Staudt, A. Lochbaum, et al., "Embedded fiber optic sensing for accurate state estimation in advanced battery management systems," MRS Proceedings, vol. 1681, pp. mrss14-1681-q01-03, June 2014.
- [179] J. Lee, F. Wu, W. Zhao, M. Ghaffari, L. Liao, and D. Siegel, "Prognostics and health management design for rotary machinery systems—Reviews, methodology and applications," *Mech. Syst. Sig. Process.*, vol. 42, no. 1-2, pp. 314-334, 2014.